A parameter-free total Lagrangian smooth particle hydrodynamics algorithm applied to problems with free surfaces

Kenny W. Q. Low¹ · Chun Hean Lee² · Antonio J. Gil³ · Jibran Haider⁴ · Javier Bonet⁵

Received: 23 August 2020 / Accepted: 28 October 2020 $\ensuremath{\textcircled{}}$ The Author(s) 2021

Abstract

This paper presents a new Smooth Particle Hydrodynamics computational framework for the solution of inviscid free surface flow problems. The formulation is based on the Total Lagrangian description of a system of first-order conservation laws written in terms of the linear momentum and the Jacobian of the deformation. One of the aims of this paper is to explore the use of Total Lagrangian description in the case of large deformations but without topological changes. In this case, the evaluation of spatial integrals is carried out with respect to the initial undeformed configuration, yielding an extremely efficient formulation where the need for continuous particle neighbouring search is completely circumvented. To guarantee stability from the SPH discretisation point of view, consistently derived Riemann-based numerical dissipation is suitably introduced where global numerical entropy production is demonstrated via a novel technique in terms of the time rate of the Hamiltonian of the system. Since the kernel derivatives presented in this work are fixed in the reference configuration, the non-physical clumping mechanism is completely removed. To fulfil conservation of the global angular momentum, a posteriori (least-squares) projection procedure is introduced. Finally, a wide spectrum of dedicated prototype problems is thoroughly examined. Through these tests, the SPH methodology overcomes by construction a number of persistent numerical drawbacks (e.g. hour-glassing, pressure instability, global conservation and/or completeness issues) commonly found in SPH literature, without resorting to the use of any ad-hoc user-defined artificial stabilisation parameters. Crucially, the overall SPH algorithm yields equal second order of convergence for both velocities and pressure.

Keywords Conservation laws · SPH · Hamiltonian · Stabilisation · Instability · Fluid

Chun Hean Lee chunhean.lee@glasgow.ac.uk

> Kenny W. Q. Low k.w.q.low@swansea.ac.uk Antonio J. Gil a.j.gil@swansea.ac.uk

- ¹ ASTUTE 2020, Swansea University, Bay Campus, Swansea SA1 8EN, UK
- ² Glasgow Computational Engineering Centre, James Watt School of Engineering, University of Glasgow, Glasgow G12 8QQ, UK
- ³ Zienkiewicz Centre for Computational Engineering, College of Engineering, Swansea University, Bay Campus, Swansea SA1 8EN, UK
- ⁴ Cambridge Flow Solutions Ltd., Cambridge Science Park, Cambridge CB4 0GZ, UK
- ⁵ University of Greenwich, London SE10 9LS, UK

1 Introduction

Modelling interfacial flows characterised by complex free surface patterns is regarded as one of the most challenging topics in the field of Computational Fluid Dynamics [1–3]. Efforts in developing a Lagrangian meshfree method to accurately capture the complex evolution of the flow interfaces still remains a challenging problem [4,5].

The Smoothed Particle Hydrodynamics (SPH) method, originally developed by Lucy [6] and Gingold and Monaghan [7] for astrophysical and cosmological applications, has since been explored in a wide range of engineering applications, namely shock hydrodynamics [8], free surface flows [9–12], multi-phase flows [13–18], rheology [19–21], magneto-hydrodynamics [22] and solid mechanics [23–31]. One attractive feature of SPH is its mesh free nature, thus not requiring the use of an underlying (or background) mesh. The absence of mesh and the calculation of the interactions among particles based exclusively on their separation allow



ease of computation for large deformation problems. For its low computational cost, reasonable accuracy and stability, as well as its ability to handle extremely large distortions [32–37], the SPH methodology has been shown to be very competitive [4].

However, as it is well established, the traditional SPH method presents a number of numerical drawbacks, namely: (1) non-physical clumping of particles due to tensile instability [38–41] when using a non-Lagrangian description of the problem, (2) numerical errors near the boundaries due to reduced compact support [42,43], (3) zero-energy modes due to rank-deficiency inherent to the use of a nodal integration approach [44], (4) pressure instabilities when considering weakly compressible fluids [45,46] and (5) reduced order of convergence for volumetric strain and pressure [47–50].

In the last two decades, substantial effort has been devoted to address above deficiencies and thus enhance the robustness of the standard SPH method. Specifically, corrections have been applied in order to ensure reproducibility of complete polynomials in finite domains, as well as to pass the patch test. Without being exhaustive, Johnson et al. [25,51] proposed a normalised smoothing technique; Chen et al. [52,53] introduced corrected kernel approximations on the basis of a Taylor series expansion; Bonet et al. [54-56] introduced corrections in the kernel functions and in their derivatives; Liu et al. [27] presented an in-depth discussion about reproducibility properties of the SPH method. Other notable modifications to the SPH method include the moving least-square particle hydrodynamics [57,58] and reproducing kernel particle method [59,60]. Problems associated with the stability and convergence of meshfree methods [24,43,61] were also reported in [61].

However, above enhanced (also known as modified or corrected) SPH methodologies still suffer from one or more of the following spurious mechanisms, namely zero-energy modes, pressure oscillations and the possible development of long-term instabilities. These numerical artefacts can be alleviated through the use of numerical dissipation mechanisms as follows. First, for the treatment of zero-energy modes, one popular option is the incorporation of ad-hoc artificial viscosity stabilisation, as proposed by Monaghan [8,9,34], to the linear momentum equation. Even though the viscosity term requires several user defined (problem dependent) stabilisation parameters, this approach is still being widely accepted and implemented in many open source codes [10,62,63]. Morris and Monaghan [64] further improved the accuracy of the stabilisation by introducing an appropriate limiter. Second, for the treatment of pressure instabilities, Monaghan [9,65] introduced a smoothing procedure for the density update, named "X-SPH", by penalising the difference between the particle velocity and the average velocity in its neighbourhood. Another a posteriori approach [10,14,62,63] is the re-initialisation of the density field via a corrected (or re-normalised) kernel function after a certain number of time steps. Unfortunately, as reported in [11], both of these techniques may potentially lead to excessive dissipation, which is clearly undesirable in practice. In parallel, Molteni and Colagrossi [45] introduced another ad-hoc approach, named δ -SPH method [66,67], by incorporating a (Laplacian based) diffusive term to the continuity equation. Further improvement has been made by Sun et al. [68] whereby a particle shifting procedure is combined. On another front, several interesting attempts have also been reported at aiming to solve the instability issues via the use of a physically based Riemann solver [69–74,74–77]. Despite the tremendous development in the field, the ab initio stability of SPH schemes is still an open problem.

In the current paper, a parameter-free Total Lagrangian SPH computational framework is presented with particular attention paid to the analysis of fluid flow problems without significant topological changes on surface (to be explored in a subsequent publication). A first-order system of hyperbolic equations¹ is presented in terms of the linear momentum pand the Jacobian J of the deformation. The mixed $\{p, J\}$ system ensures the pressure to converge at the same spatial rate than the velocities and displacements, advantageous in those scenarios (especially when considering inviscid fluid flow problems) where the pressure plays a dominant role. One contribution of this paper is the reformulation of the Updated Lagrangian version of the inviscid fluid equations typically used in SPH in a Total Lagrangian description. In this case, the evaluation of spatial integrals is carried out with respect to the initial undeformed configuration, yielding an extremely efficient formulation where the need for continuous particle neighbouring search is avoided altogether. One of the objectives of this paper is to lay the foundation for seeking a new SPH approach with the final aim to handle in the near future violent free surface flows typically accompanied with large topological changes.

A crucial aspect that requires special attention is that of the stability of the SPH formulation. Specifically, a Riemann based (upwinding) approach is pursued where a consistently derived numerical stabilisation is carefully introduced ensuring global production of numerical entropy. The latter is demonstrated by the monitoring of the so-called Hamiltonian of the system [88]. For completeness, a global a posteriori angular momentum projection procedure is also presented with the aim to preserve the angular momentum of the system. On the shown test problems, the proposed SPH algorithm is shown to effectively eliminate the appearance of spurious hourglass-like modes and pressure instabilities.

The paper is organised as follows. Section 2 summarises the system of first-order conservation laws $\{p, J\}$ for inviscid

¹ Similar first-order conservation laws have been recently exploited in the field of solid mechanics [29–31,78–87].



Fig. 1 Motion of a deformable continuum domain

fluid flow problems used in this paper. Section 3 presents the weak variational statements of the problem as well the second law of thermodynamics written in terms of the so-called Hamiltonian [86,89,90]. Section 4 presents the Smooth Particle Hydrodynamics discretisation numerical scheme where special attention is paid to the Riemann based (upwinding) numerical dissipation employed. A proof of entropy production is included as a necessary condition for stability at the semi-discrete level. Section 5 describes the explicit Total Variation Diminishing Runge-Kutta time integrator used for temporal discretisation. Section 6 summarises the flowchart of the proposed SPH methodology. In Sect. 7, an extensive set of challenging numerical examples is examined to assess the performance of the proposed algorithm. Finally, Sect. 8 presents some concluding remarks and current directions of research.

2 Total Lagrangian first-order conservation laws

2.1 Kinematics and conservation of volume map

Consider the three dimensional deformation of a continuum moving from its initial (material) configuration occupying a volume Ω_0 , of boundary $\partial \Omega_0$ with outward unit normal N, into a current configuration at time t occupying a volume $\Omega(t)$ of boundary $\partial \Omega(t)$ with outward unit normal n, see Fig. 1. The standard notation and definitions for the deformation gradient tensor F (used to map differential fibre elements $d\mathbf{x} = F dX$) and its determinant J (used to map differential volume elements $d\Omega = J d\Omega_0$) are introduced

$$F = \frac{\partial x}{\partial X} = \nabla_0 x; \qquad J = \det F, \tag{1}$$

where x represents the current position of a particle originally at X and ∇_0 denotes the gradient with respect to the material configuration. Similarly, material differential area

vectors dA (co-linear with N) are mapped to current differential area vectors da (co-linear with n) through the co-factor H of the deformation, which is related to the deformation gradient tensor F via the Nanson's rule [86]

$$H = J F^{-T}.$$
 (2)

An alternative way to Eq. (1b) to compute the Jacobian of the deformation is possible via an integral conservation law [29-31,80,84-86] as follows,

$$\frac{\mathrm{d}}{\mathrm{d}t} \int_{\Omega_0} J \,\mathrm{d}\Omega_0 = \int_{\partial\Omega_0} \left(\boldsymbol{H}^{\mathrm{T}} \boldsymbol{v} \right) \cdot N \mathrm{d}A,\tag{3}$$

where $v = \frac{dx}{dt}$ is the velocity field. The equivalent local conservation law and associated jump condition across a discontinuity surface with normal *N*, propagating with speed *U* in the reference space [78,83,84], are

$$\frac{\partial J}{\partial t} - \mathrm{DIV}\left(\boldsymbol{H}^{\mathrm{T}}\boldsymbol{v}\right) = 0; \qquad U[\![J]\!] = -[\![\boldsymbol{v}]\!] \cdot \left(\boldsymbol{H}^{\mathrm{Ave}}\boldsymbol{N}\right). \tag{4}$$

In this expression, DIV represents the material divergence operator, $\boldsymbol{H}^{Ave} = \frac{1}{2} (\boldsymbol{H}^- + \boldsymbol{H}^+)$ represents an average state of the co-factor of the deformation between the left and right states across the discontinuity surface and $[\![\cdot]\!] = [\cdot]^+ - [\cdot]^-$ represents the jump operator.

2.2 Conservation of linear momentum

The conservation of linear momentum per unit undeformed volume $p = \rho_0 v$ (with ρ_0 the material density of the continuum) [29–31,78–87] is established for any arbitrary Lagrangian material volume Ω_0 by

$$\frac{\mathrm{d}}{\mathrm{d}t} \int_{\Omega_0} \boldsymbol{p} \,\mathrm{d}\Omega_0 = \int_{\partial\Omega_0} \boldsymbol{t} \,\mathrm{d}A + \int_{\Omega_0} \boldsymbol{f}_0 \,\mathrm{d}\Omega_0, \tag{5}$$

where f_0 is a body force per unit of undeformed volume and t = PN is the traction vector associated with the material outward unit normal surface vector N with P being the first Piola-Kirchhoff stress tensor. The equivalent local equilibrium equation and the corresponding jump condition across a discontinuity [78,82–84] can be written as

$$\frac{\partial \boldsymbol{p}}{\partial t} - \text{DIV}\boldsymbol{P} = \boldsymbol{f}_0; \qquad U\rho_0[\![\boldsymbol{v}]\!] = -[\![\boldsymbol{P}]\!]N. \tag{6}$$

2.3 Combined equations

Combining expressions (3) and (5), above global conservation laws can now be summarised in a concise manner as

$$\frac{\mathrm{d}}{\mathrm{d}t} \int_{\Omega_0} \mathcal{U} \,\mathrm{d}\Omega_0 = -\int_{\partial\Omega_0} \mathcal{F}_N \,\mathrm{d}A + \int_{\Omega_0} \mathcal{S} \,\mathrm{d}\Omega_0, \tag{7}$$

D Springer

representing a system of Total Lagrangian conservation laws where \mathcal{U} denotes the set of conservation variables, \mathcal{S} the source term and \mathcal{F}_N the surface flux vector, described as follows

$$\mathcal{U} = \begin{bmatrix} \boldsymbol{p} \\ J \end{bmatrix}; \quad \mathcal{F}_N = -\begin{bmatrix} \boldsymbol{t} \\ \frac{1}{\rho_0} \boldsymbol{p} \cdot (\boldsymbol{H}N) \end{bmatrix}; \quad \mathcal{S} = \begin{bmatrix} \boldsymbol{f}_0 \\ 0 \end{bmatrix},$$
(8)

where H is the cofactor of the deformation previously defined in (2). The computation of H is carried out from the use of Jin (4) and the computation of the deformation gradient tensor (1) obtained from the time integration of the velocity field vdeduced from (6).

For closure, above system (8) requires the introduction of an appropriate "elastic fluid" type of constitutive law (also known as polytropic equation of state [34,54]), relating the stress tensor P with the Jacobian of the deformation J. In this specific case, a strain energy Ψ per unit undeformed volume is introduced as

$$\Psi(J) = \begin{cases} \kappa \left(J - 1 - \frac{J^{1-\gamma} - 1}{1-\gamma}\right), \ \gamma > 1\\ \kappa (J - 1 - \ln J), \qquad \gamma = 1 \end{cases}$$
(9)

with $\{\gamma, \kappa\} > 0$ the positive material constants.² Thus, the first Piola-Kirchhoff stress tensor *P* can be expressed as

$$\boldsymbol{P} = -p\boldsymbol{H}; \qquad \boldsymbol{p} = -\frac{\mathrm{d}\Psi}{\mathrm{d}J} = \kappa \left[\left(\frac{1}{J}\right)^{\gamma} - 1 \right], \tag{10}$$

where a positive value of the pressure p in above equation indicates compression. So far only an elastic volumetric contribution (10) is considered. General materials however will also exhibit additional inelastic effects P^{v} , where the combined stress becomes

$$\boldsymbol{P} = -p\boldsymbol{H} + \boldsymbol{P}^{\boldsymbol{v}}.\tag{11}$$

In the case of fluids this is due to viscosity whereas for solids more complex constitutive equations possibly involving plasticity [31] can be required. In the current paper, we consider the case of perfect fluid, that is $P^v = 0$. Physically, this implies that no shear/deviatoric stresses, viscosity or heat conduction are taken into account.

Importantly, with the aid of (9), hyperbolicity of system (8) can be demonstrated (see "Appendix") ensuring the existence of real wave speeds at any state of deformation. Finally, for the complete definition of the initial boundary value problem, initial and boundary (essential and natural) conditions must also be specified as appropriate.

Remark 1 An alternative non-conservative form for J (4a) is achieved by introducing the Piola identity, namely DIVH = 0, into Eq. (4a) to give

$$\frac{\partial J}{\partial t} - \boldsymbol{H} : \boldsymbol{\nabla}_0 \boldsymbol{v} = 0.$$
⁽¹²⁾

3 Weak form statements

In general, a standard weak variational statement for the system (8) is established by multiplying the local differential equations \mathcal{U} (8) with their appropriate work conjugate virtual fields $\delta \mathcal{V}$, and integrating over the reference domain Ω_0 of the body, to give

$$\int_{\Omega_0} \delta \boldsymbol{\mathcal{V}} \bullet \frac{\partial \boldsymbol{\mathcal{U}}}{\partial t} \, \mathrm{d}\Omega_0 = \int_{\Omega_0} \delta \boldsymbol{\mathcal{V}} \bullet \boldsymbol{\mathcal{S}} \, \mathrm{d}\Omega_0 + \int_{\Omega_0} \boldsymbol{\mathcal{F}}_I \bullet \frac{\partial \delta \boldsymbol{\mathcal{V}}}{\partial X_I} \, \mathrm{d}\Omega_0$$
$$- \int_{\partial \Omega_0} \delta \boldsymbol{\mathcal{V}} \bullet \boldsymbol{\mathcal{F}}_N \, \mathrm{d}A, \tag{13}$$

where the normal fluxes $\mathcal{F}_N = \mathcal{F}_I N_I$ with N_I being the material outward normal in *I*-th material direction and the symbol \bullet is used to denote the inner (dual) product of work conjugate pairs.

In order to give a proper physical meaning to the conjugate virtual fields $\delta \mathcal{V}$ and pave the way for the proof of numerical entropy production presented in a subsequent section, it is useful to introduce the notion of Hamiltonian $\mathcal{H} = \mathcal{H}(X, t)$ [86,88]. This indeed can be understood as a generalised convex entropy function of the system of conservation laws (8). Specifically, the Hamiltonian \mathcal{H} is defined by

$$\mathcal{H}(\boldsymbol{X},t) = \hat{\mathcal{H}}(\boldsymbol{\mathcal{U}}) = \frac{1}{2\rho_0} \boldsymbol{p} \cdot \boldsymbol{p} + \Psi(J), \qquad (14)$$

which represents the summation of the kinetic energy (i.e. the first term on the right hand side of (14)) and Helmholtz free energy Ψ per unit of undeformed volume. Note here that $\mathcal{H}(X, t)$ and $\hat{\mathcal{H}}(\mathcal{U})$ represent alternative functional representations of the same quantity. With expression (14) at hand, the work conjugates \mathcal{V} can then be obtained by taking simple differentiation to give

$$\mathcal{V} = \frac{\partial \hat{\mathcal{H}}}{\partial \mathcal{U}} = \begin{bmatrix} \frac{\partial \hat{\mathcal{H}}}{\partial p} \\ \frac{\partial \hat{\mathcal{H}}}{\partial J} \end{bmatrix} = \begin{bmatrix} \mathbf{v} \\ -p \end{bmatrix}.$$
(15)

Finally, upon substitution of (15) into (13), individual variational statement for the linear momentum p and the Jacobian

² Crucially, differentiating Eq. (9) twice with respect to *J* implies the convexity of $\Psi(J)$, that is $\frac{d^2\Psi(J)}{dJ^2} \ge 0$ for all $\gamma \ge 1$.

of the deformation *J* are

$$\int_{\Omega_0} \delta \boldsymbol{v} \cdot \frac{\partial \boldsymbol{p}}{\partial t} \, \mathrm{d}\Omega_0 = -\int_{\Omega_0} \boldsymbol{P} : \boldsymbol{\nabla}_0 \delta \boldsymbol{v} \, \mathrm{d}\Omega_0 + \int_{\Omega_0} \delta \boldsymbol{v} \cdot \boldsymbol{f}_0 \, \mathrm{d}\Omega_0 + \int_{\partial\Omega_0} \delta \boldsymbol{v} \cdot \boldsymbol{t}_B \, \mathrm{d}A; \int_{\Omega_0} \delta \boldsymbol{p} \frac{\partial J}{\partial t} \, \mathrm{d}\Omega_0 = \int_{\Omega_0} \delta \boldsymbol{p} \left(\boldsymbol{H} : \boldsymbol{\nabla}_0 \boldsymbol{v} \right) \, \mathrm{d}\Omega_0.$$
(16)

Here, both δv and δp represent the virtual velocity and pressure fields, respectively.

3.1 Second law of thermodynamics

It is instructive to revisit the second law of thermodynamics when written in terms of the Hamiltonian. The time rate of the volume integral of the Hamiltonian is obtained as follows

$$\frac{\mathrm{d}}{\mathrm{d}t} \int_{\Omega_0} \mathcal{H} \,\mathrm{d}\Omega_0 = \int_{\Omega_0} \mathcal{V} \bullet \frac{\partial \mathcal{U}}{\partial t} \,\mathrm{d}\Omega_0$$

$$= \int_{\Omega_0} \left(\boldsymbol{v} \cdot \frac{\partial \boldsymbol{p}}{\partial t} - \boldsymbol{p} \frac{\partial J}{\partial t} \right) \,\mathrm{d}\Omega_0$$

$$= \int_{\Omega_0} \left(\boldsymbol{v} \cdot \frac{\partial \boldsymbol{p}}{\partial t} - \boldsymbol{p} \boldsymbol{H} : \boldsymbol{\nabla}_0 \boldsymbol{v} \right) \,\mathrm{d}\Omega_0$$

$$= \int_{\Omega_0} \left(\boldsymbol{v} \cdot \frac{\partial \boldsymbol{p}}{\partial t} + (\boldsymbol{P} - \boldsymbol{P}^v) : \boldsymbol{\nabla}_0 \boldsymbol{v} \right) \,\mathrm{d}\Omega_0$$
(17)

where, in the first line of (17), use has been made of the conjugacy of the fields (15). In addition, Eqs. (12) and (11) have been substituted in the second and third lines of (17), respectively. Subsequently, we can substitute the linear momentum Eqs. (6) into (17) to give

$$\frac{\mathrm{d}}{\mathrm{d}t} \int_{\Omega_0} \mathcal{H} \,\mathrm{d}\Omega_0 = \int_{\Omega_0} \left(\boldsymbol{v} \cdot \boldsymbol{f}_0 + \boldsymbol{v} \cdot \mathrm{DIV} \boldsymbol{P} + \boldsymbol{P} : \boldsymbol{\nabla}_0 \boldsymbol{v} \right) \,\mathrm{d}\Omega_0$$
$$- \int_{\Omega_0} \boldsymbol{P}^v : \boldsymbol{\nabla}_0 \boldsymbol{v} \,\mathrm{d}\Omega_0.$$
(18)

Recalling that $\boldsymbol{v} \cdot \text{DIV}\boldsymbol{P} + \boldsymbol{P}$: $\nabla_0 \boldsymbol{v} = \text{DIV}(\boldsymbol{P}^T \boldsymbol{v})$, above equation reduces to

$$\frac{\mathrm{d}}{\mathrm{d}t} \int_{\Omega_0} \mathcal{H} \,\mathrm{d}\Omega_0 = \int_{\Omega_0} \left(\boldsymbol{v} \cdot \boldsymbol{f}_0 + \mathrm{DIV} \left(\boldsymbol{P}^{\mathrm{T}} \boldsymbol{v} \right) \right) \,\mathrm{d}\Omega_0$$
$$- \int_{\Omega_0} \boldsymbol{P}^{\boldsymbol{v}} : \boldsymbol{\nabla}_0 \boldsymbol{v} \,\mathrm{d}\Omega_0.$$
(19)

By performing integration by parts of the DIV term in Eq. (19) and after some re-arrangement, it yields

$$\frac{\mathrm{d}}{\mathrm{d}t} \int_{\Omega_0} \mathcal{H} \,\mathrm{d}\Omega_0 - \dot{\Pi}_{\mathrm{ext}} = -\int_{\Omega_0} \boldsymbol{P}^{\boldsymbol{\nu}} : \boldsymbol{\nabla}_0 \boldsymbol{\upsilon} \,\mathrm{d}\Omega_0, \tag{20}$$

where $\dot{\Pi}_{ext}$ denotes the power introduced by external forces, defined as

$$\dot{\Pi}_{\text{ext}} = \int_{\Omega_0} \boldsymbol{v} \cdot \boldsymbol{f}_0 \, \mathrm{d}\Omega_0 + \int_{\partial\Omega_0} \boldsymbol{v} \cdot \boldsymbol{t}_B \, \mathrm{d}A. \tag{21}$$

In expression (20), the term on the right hand side represents possible physical dissipation and is always non-positive (namely, $-\mathbf{P}^{v}$: $\nabla_{0} v \leq 0$) and, consequently, Eq. (20) can be transformed into the following inequality

$$\frac{\mathrm{d}}{\mathrm{d}t} \int_{\Omega_0} \mathcal{H} \,\mathrm{d}\Omega_0 - \dot{\Pi}_{\mathrm{ext}} \le 0, \tag{22}$$

which represents a valid expression for the second law of thermodynamics [91]. Satisfaction of inequality (22) is a necessary ab initio condition to ensure stability, otherwise referred to as the Coleman–Noll procedure [91].

In the specific case of an isolated system (i.e. $\dot{\Pi}_{ext} = 0$), inequality (22) reduces to

$$\frac{\mathrm{d}}{\mathrm{d}t} \int_{\Omega_0} \mathcal{H} \,\mathrm{d}\Omega_0 \le 0. \tag{23}$$

This implies that for an isolated system, the decrease in the Hamiltonian is intrinsically related to the dissipation introduced by any inelastic (i.e. viscous) effects. This key concept will be further exploited in Sect. 4.2 at a semi-discrete level.

4 Spatial discretisation

For the case of corrected SPH methods [54], both the problem variables \mathcal{U} and the conjugate pairs $\delta \mathcal{V}$ are in general interpolated at any given position via corrected SPH shape functions \tilde{N} (or kernel functions \tilde{W}) with a given compact support of radius 2h around every particle (see Fig. 2). Specifically, for a given position X_a , both \mathcal{U} and $\delta \mathcal{V}$ can be approximated as

$$\begin{aligned} \boldsymbol{\mathcal{U}}_{a}(t) &\approx \sum_{b \in \Lambda_{a}^{b}} \tilde{N}_{b}(\boldsymbol{X}_{a}) \boldsymbol{\mathcal{U}}_{b}(t); \qquad \delta \boldsymbol{\mathcal{V}}_{a} \approx \sum_{b \in \Lambda_{a}^{b}} \tilde{N}_{b}(\boldsymbol{X}_{a}) \delta \boldsymbol{\mathcal{V}}_{b}; \\ \tilde{N}_{b}(\boldsymbol{X}_{a}) &= \Omega_{0}^{b} \tilde{W}_{b}(\boldsymbol{X}_{a}). \end{aligned}$$
(24)

Here, Λ_a^b represents the set of neighbouring particles *b* that lie inside a sphere of a given radius around X_a , Ω_0^b represents the volume associated to particle *b* and $\mathcal{U}_b(t)$ and $\delta \mathcal{V}_b$ represent the time varying variables and their virtual conjugate



Fig. 2 The compact support of kernel approximation

pairs stored at particle *b*, respectively. In addition, $[\bullet]_a(t)$ represent the problem variables at position X_a and time *t*, that is $[\bullet]_a(t) = [\bullet](X_a, t)$. Additionally, the use of corrected SPH shape functions \tilde{N} ensures that both constant and linear functions are perfectly interpolated [54].

In line with previous work [31], the material gradient $\nabla_0(\bullet)$ of any arbitrary vector function f can be approximated through the so-called corrected pseudo area vector $\tilde{\mathbf{C}}_{ab}$ (defined on the basis of corrected kernel gradient $\tilde{\nabla}_0$ [54]), described as

$$\nabla_{0} f(X_{a}) \approx \frac{1}{\Omega_{0}^{a}} \sum_{b \in \Lambda_{a}^{b}} \frac{1}{2} \left(f_{b} - f_{a} \right) \otimes \tilde{\mathbf{C}}_{ab};$$

$$\tilde{\mathbf{C}}_{ab} := 2\Omega_{0}^{a} \nabla_{0} \tilde{\mathbf{N}}_{b}(X_{a});$$

$$\nabla_{0} \tilde{\mathbf{N}}_{b}(X_{a}) = \Omega_{0}^{b} \tilde{\nabla}_{0} W_{b}(X_{a}).$$
 (25)

Specifically, the correction technique involves modifying (uncorrected) pseudo area vector C_{ab} (or kernel gradient ∇_0) by introducing a particle based correction matrix L_a [54]

$$\tilde{\mathbf{C}}_{ab} = \mathbf{L}_a \mathbf{C}_{ab}; \quad \mathbf{C}_{ab} := 2\Omega_0^a \nabla_0 \mathbf{N}_b(\mathbf{X}_a); \quad \nabla_0 \mathbf{N}_b(\mathbf{X}_a)
= \Omega_0^b \nabla_0 W_b(\mathbf{X}_a),$$
(26)

from which L_a is explicitly evaluated as [54,55]

$$\boldsymbol{L}_{a} = \left(\sum_{b \in \Lambda_{a}^{b}} \Omega_{0}^{b} \nabla_{0} W_{b}(\boldsymbol{X}_{a}) \otimes (\boldsymbol{X}_{b} - \boldsymbol{X}_{a})\right)^{-1}.$$
 (27)

This type of kernel gradient correction [28,54] ensures the gradient of any linear field distribution is exactly evaluated. Moreover, the term $-f_a$ is added in (25) in order to ensure the

gradient vanishes for a constant (uniform) field [8]. Noticing that if no correction is applied (i.e. $L_a = I$), the corrected pseudo area vector \tilde{C}_{ab} (26) degenerates to the usual SPH uncorrected area vector C_{ab} [8].

It is worth pointing out that the correction matrices (27) applied to a pair of interacting particles *a* and *b* are in general not the same, that is $L_a \neq L_b$. This would consequently destroy the anti-symmetric condition of the gradient evaluations (i.e. $\tilde{\mathbf{C}}_{ab} \neq -\tilde{\mathbf{C}}_{ba}$) [92]. One possible option to enforce such condition consists of skew-symmetrising the area vector $\tilde{\mathbf{C}}_{ab}^{\text{Skew}}$ by taking the difference between the area vectors of particle pairs to give

$$\tilde{\mathbf{C}}_{ab}^{\text{Skew}} = \frac{1}{2} \left(\tilde{\mathbf{C}}_{ab} - \tilde{\mathbf{C}}_{ba} \right)$$
$$= L_{ab}^{\text{Ave}} \mathbf{C}_{ab}; \qquad \qquad L_{ab}^{\text{Ave}} = \frac{1}{2} (L_a + L_b).$$
(28)

4.1 Total Lagrangian SPH discrete formulation

Typically, in SPH type approaches [8,14,29-31,54], the above weak statements (16) are under-integrated by using the cloud of particles as quadrature point [8,28,41,54,55,93]. Doing this may potentially excite the instabilities (e.g. zero-energy modes and pressure instability) arising from rank-deficiency inherent to the use of particle (or nodal) integration approach. For instance, the error accumulation over time can eventually lead to the breakdown of a scheme. Additionally, in order to ensure the completeness conditions (low order polynomials are interpolated exactly) of a SPH algorithm, the type of correction technique described in expressions (24) and (25) are typically used when evaluating the integrals of (16). This correction method however is well known to destroy the desired anti-symmetric property of the pseudo area vector. In other words, pseudo area vectors between all pairwise interacting particles are no longer equal and opposite.

Now the question remains on how to construct a conservative SPH method capable of addressing persistent SPH numerical deficiencies (e.g. hour-glassing and pressure modes) described above, whilst still minimising the errors caused by the dissatisfaction of the anti-symmetric property as the result of gradient correction. To achieve this, and taking inspiration from previous work [31], a SPH method in conjunction with well established stabilisation procedure is presented. The SPH discretisation for the system {p, J} described in (7) becomes

$$\Omega_0^a \frac{\mathrm{d}\boldsymbol{p}_a}{\mathrm{d}t} = \sum_{b \in \Lambda_a^b} \boldsymbol{T}_{ab} + A_0^a \boldsymbol{t}_B^a + \Omega_0^a \boldsymbol{f}_0^a + \sum_{b \in \Lambda_a^b} \boldsymbol{\mathcal{D}}_{\boldsymbol{v}}^{ab} + \sum_{b \in \Lambda_a^b} \boldsymbol{\mathcal{D}}_{\Delta \mathbf{C}}^{ab};$$
(29a)

$$\Omega_0^a \frac{\mathrm{d}J_a}{\mathrm{d}t} = \boldsymbol{H}_a : \left[\sum_{b \in \Lambda_a^b} \frac{1}{2} \left(\boldsymbol{v}_b - \boldsymbol{v}_a \right) \otimes \tilde{\mathbf{C}}_{ab} \right] + \sum_{b \in \Lambda_a^b} \mathcal{D}_p^{ab}.$$
(29b)

In above expressions, the pairwise interacting force

$$\boldsymbol{T}_{ab} = \frac{1}{2} \left[\boldsymbol{P}_a \tilde{\mathbf{C}}_{ab} - \boldsymbol{P}_b \tilde{\mathbf{C}}_{ba} \right],\tag{30}$$

clearly satisfies local conservation by construction, namely $T_{ab} = -T_{ba}$. Moreover, the discrete cofactor $H_a = J_a F_a^{-T}$ (2) is evaluated using the discrete deformation gradient F_a accurately approximated as

$$\boldsymbol{F}_{a} = \frac{1}{\Omega_{0}^{a}} \sum_{b \in \Lambda_{a}^{b}} \frac{1}{2} \left(\boldsymbol{x}_{b} - \boldsymbol{x}_{a} \right) \otimes \tilde{\mathbf{C}}_{ab}, \tag{31}$$

where x_a represent the current nodal geometry at particle *a* obtained from time integration of the velocity v_a .

Finally, the remaining terms to be defined in Eq. (29a, b)the so-called pairwise stabilisation are terms $\{\mathcal{D}_{v}^{ab}, \mathcal{D}_{p}^{ab}, \mathcal{D}_{\Delta C}^{ab}\}$. In particular, the term \mathcal{D}_{ab}^{v} in (29a) addresses spurious zero-energy (hourglass-like [33]) modes due to rank-deficiency, whereas the term \mathcal{D}_{ab}^{p} (29b) removes pressure instabilities in near incompressibility regimes [80]. Another challenging aspect when designing a SPH numerical scheme is the ability to control errors arising from the violation of the skew-symmetric nature due to the gradient correction. One viable option is the introduction of an extra stabilisation term $\mathcal{D}^{ab}_{\Lambda C}$ into expression (29a), with the objective to penalise the difference between the usual uncorrected pseudo area vector \mathbf{C}_{ab} and $\tilde{\mathbf{C}}_{ab}^{\text{Skew}}$ (28). Mathematically, this is described as $\mathbf{C}_{ab} - \tilde{\mathbf{C}}_{ab}^{\text{Skew}}$.

To ensure the global conservation of the SPH discretised Eq. (29), and following the style of (30), we choose to strongly enforce local conservation for the stabilisations involved, in such a way that $\mathcal{D}_{v}^{ab} = -\mathcal{D}_{v}^{ba}$, $\mathcal{D}_{p}^{ab} = -\mathcal{D}_{p}^{ba}$ and $\mathcal{D}_{\Delta C}^{ab} = -\mathcal{D}_{\Delta C}^{ba}$. Definition of these terms follows via the semi-discrete version of the Colemann-Noll procedure [94] in order to ensure the production of numerical entropy. This will be discussed in the following section.

4.2 Numerical entropy production

In this section, we present a procedure to obtain consistent and locally conservative stabilisation terms by utilising the concept of the rate of entropy production [95–97], which can be understood as a semi-discrete version of the classical Coleman–Noll procedure. Specifically, the semi-discrete counterpart of (17) is

$$\sum_{a} \Omega_{0}^{a} \frac{\mathrm{d}\mathcal{H}_{a}}{\mathrm{d}t} = \sum_{a} \Omega_{0}^{a} \mathcal{V}_{a} \cdot \frac{\mathrm{d}\mathcal{U}_{a}}{\mathrm{d}t}$$
$$= \sum_{a} \Omega_{0}^{a} \left[\mathbf{v}_{a} \cdot \frac{\mathrm{d}\mathbf{p}_{a}}{\mathrm{d}t} - p_{a} \frac{\mathrm{d}J_{a}}{\mathrm{d}t} \right].$$
(32)

Subsequently, we can substitute the linear momentum equation (29a) and the Jacobian equation (29b) into (32), and after some algebra, gives

$$\sum_{a} \Omega_{0}^{a} \frac{\mathrm{d}\mathcal{H}_{a}}{\mathrm{d}t} = \sum_{a} \sum_{b \in \Lambda_{a}^{b}} \frac{1}{2} \Big[\Big(\boldsymbol{P}_{a} \tilde{\mathbf{C}}_{ab} \Big) \cdot \boldsymbol{v}_{b} \\ - \Big(\boldsymbol{P}_{b} \tilde{\mathbf{C}}_{ba} \Big) \cdot \boldsymbol{v}_{a} \Big] + \dot{\Pi}_{\text{ext}} \\ + \underbrace{\sum_{a} \sum_{b \in \Lambda_{a}^{b}} \Big(\boldsymbol{v}_{a} \cdot \boldsymbol{\mathcal{D}}_{\boldsymbol{v}}^{ab} - p_{a} \mathcal{D}_{p}^{ab} + \boldsymbol{v}_{a} \cdot \boldsymbol{\mathcal{D}}_{\Delta C}^{ab} \Big)}_{-\mathcal{D}_{\text{total}}}.$$
(33)

Here, $\dot{\Pi}_{ext}$ denotes the semi-discrete power introduced by external forces expressed as

$$\dot{\Pi}_{\text{ext}} = \sum_{a} \left(\Omega_0^a \boldsymbol{v}_a \cdot \boldsymbol{f}_0^a + A_0^a \boldsymbol{v}_B^a \cdot \boldsymbol{t}_B^a \right).$$
(34)

Additionally, due to the anti-symmetric nature of the first term in the first line of (33), we can conclude that this term cancel and thus (33) reduces to

$$\sum_{a} \Omega_0^a \frac{\mathrm{d}\mathcal{H}_a}{\mathrm{d}t} - \dot{\Pi}_{\mathrm{ext}} = -\mathcal{D}_{\mathrm{total}}.$$
(35)

It is now the objective to demonstrate that the term on the right hand side of (35) is always non-positive, that is $\mathcal{D}_{total} \geq 0$ (to be in agreement with (22)). Specifically,

$$\mathcal{D}_{\text{total}} = -\sum_{a} \sum_{b \in \Lambda_{a}^{b}} \left(\boldsymbol{v}_{a} \cdot \boldsymbol{\mathcal{D}}_{\boldsymbol{v}}^{ab} - p_{a} \mathcal{D}_{p}^{ab} + \boldsymbol{v}_{a} \cdot \boldsymbol{\mathcal{D}}_{\Delta \mathbf{C}}^{ab} \right)$$
$$= -\sum_{a} \sum_{b \in \Lambda_{a}^{b}} \left(\boldsymbol{v}_{b} \cdot \boldsymbol{\mathcal{D}}_{\boldsymbol{v}}^{ba} - p_{b} \mathcal{D}_{p}^{ba} + \boldsymbol{v}_{b} \cdot \boldsymbol{\mathcal{D}}_{\Delta \mathbf{C}}^{ba} \right)$$
(36)

Adding the first line and the second line of the equation above, and again noting the anti-symmetric nature of the stabilisation terms (e.g. $\mathcal{D}_{v}^{ab} = -\mathcal{D}_{v}^{ba}$, $\mathcal{D}_{p}^{ab} = -\mathcal{D}_{p}^{ba}$, $\mathcal{D}_{\Delta C}^{ab} = -\mathcal{D}_{\Delta C}^{ba}$), an alternative expression for $\mathcal{D}_{\text{total}}$ is

$$\mathcal{D}_{\text{total}} = \frac{1}{2} \sum_{a} \sum_{b \in \Lambda_a^b} \mathcal{D}_{\text{total}}^{ab}$$
(37)

$$\mathcal{D}_{\text{total}}^{ab} = (\boldsymbol{v}_b - \boldsymbol{v}_a) \cdot \mathcal{D}_{\boldsymbol{v}}^{ab} - (p_b - p_a) \mathcal{D}_p^{ab} + (\boldsymbol{v}_b - \boldsymbol{v}_a) \cdot \mathcal{D}_{\Delta \mathbf{C}}^{ab}.$$
(38)

Dissipation terms $\{\mathcal{D}_{v}^{ab}, \mathcal{D}_{p}^{ab}, \mathcal{D}_{\Delta C}^{ab}\}$ remain to be defined in order to ensure non-negative entropy production, namely, $\mathcal{D}_{\text{total}} > 0$. In this work, the term $\mathcal{D}_{\Delta C}^{ab}$ is defined as

$$\mathcal{D}_{\Delta \mathbf{C}}^{ab} = \frac{1}{2} \left(\mathbf{P}_a + \mathbf{P}_b \right) \left(\mathbf{C}_{ab} - \tilde{\mathbf{C}}_{ab}^{\text{Skew}} \right), \tag{39}$$

which is designed in order to penalise the violation of the skew-symmetric nature of the gradient kernel correction. Note that the above definition of $\mathcal{D}^{ab}_{\Delta C}$ ensures the skew symmetric nature of this term. As for the remaining stabilisation terms, we propose the following definitions

$$\mathcal{D}_{\boldsymbol{v}}^{ab} = S_{ab}^{\boldsymbol{v}} \left(\boldsymbol{v}_b - \boldsymbol{v}_a \right);$$

$$\mathcal{D}_p^{ab} = -S_{ab}^p \left(p_b - p_a \right),$$
(40)

where $S_{ab}^{v} = S_{ba}^{v}$ is a positive definite stabilisation matrix and $S_{ab}^{p} = S_{ba}^{p} > 0$. Notice how the dissipation terms $\{\mathcal{D}_{v}^{ab}, \mathcal{D}_{p}^{ab}\}$ defined in (40) are skew symmetric and proportional to the difference (or jump) in velocity $(v_{b} - v_{a})$ and pressure $(p_{b} - p_{a})$ between pairwise particles, typical of Riemann solver based upwinding terms [78]. Moreover, with these definitions for $\{\mathcal{D}_{v}^{ab}, \mathcal{D}_{p}^{ab}\}$ (40), the first two terms on the right hand side of (38) are always strictly positive, thus counterbalancing possible negative values taken by the third term. In this paper, the dissipation terms used are specifically chosen as

$$\boldsymbol{S}_{ab}^{\boldsymbol{v}} = \frac{1}{2} \rho_{R,ab}^{\text{Ave}} \| \tilde{\boldsymbol{C}}_{ab}^{\text{Skew}} \| c_{p,ab}^{\text{Ave}} \boldsymbol{I}; \qquad \boldsymbol{S}_{ab}^{p} = \frac{\tilde{\boldsymbol{c}}_{ab}^{\text{Skew}} \cdot \tilde{\boldsymbol{c}}_{ab}^{\text{Skew}}}{2\rho_{R,ab}^{\text{Ave}} c_{p,ab}^{\text{Ave}} \| \tilde{\boldsymbol{C}}_{ab}^{\text{Skew}} \|},$$
(41)

with

$$\rho_{R,ab}^{\text{Ave}} = \frac{1}{2} \left(\rho_{R,a} + \rho_{R,b} \right); \qquad c_{p,ab}^{\text{Ave}} = \frac{1}{2} \left(c_{p,a} + c_{p,b} \right),$$
(42)

respectively. Here, $\rho_{R,a}$ denotes the material density evaluated at particle *a* and $c_{p,a}$ represents the pressure wave speed (58) computed at particle *a* and $\tilde{\mathbf{c}}_{ab}^{\text{Skew}} = \frac{1}{2} \left(\boldsymbol{H}_a \tilde{\mathbf{C}}_{ab} - \boldsymbol{H}_b \tilde{\mathbf{C}}_{ba} \right)$.

Remark 2 Notice that it is always possible to ensure ab initio non-negative entropy production by selecting an appropriate numerical value for the wave speed, possibly different to

 $c_{p,ab}^{\text{Ave}}$ in (42). For instance, selecting $S_{ab}^{p} = 0$, the pairwise numerical dissipation becomes

$$\mathcal{D}_{\text{total}}^{ab} = (\boldsymbol{v}_b - \boldsymbol{v}_a) \cdot \mathcal{D}_{\boldsymbol{v}}^{ab} + (\boldsymbol{v}_b - \boldsymbol{v}_a) \cdot \mathcal{D}_{\Delta \mathbf{C}}^{ab}$$
$$= \frac{1}{2} \rho_{R,ab}^{\text{Ave}} \tilde{c}_{p,ab} \|\tilde{\mathbf{C}}_{ab}^{\text{Skew}}\| \|\boldsymbol{v}_b - \boldsymbol{v}_a\|^2$$
$$+ (\boldsymbol{v}_b - \boldsymbol{v}_a) \cdot \mathcal{D}_{\Delta \mathbf{C}}^{ab} \ge 0,$$
(43)

where $\tilde{c}_{p,ab}$ is an appropriate wave speed which can be selected as

$$\tilde{c}_{p,ab} = \max\left\{c_{p,ab}^{\text{Ave}}, \frac{-2(\boldsymbol{v}_b - \boldsymbol{v}_a) \cdot \boldsymbol{\mathcal{D}}_{\Delta \mathbf{C}}^{ab}}{\rho_{R,ab}^{\text{Ave}} \|\tilde{\mathbf{C}}_{ab}^{\text{Skew}}\|\|\boldsymbol{v}_b - \boldsymbol{v}_a\|^2}\right\}, \quad (44)$$

thus ensuring positive entropy production.

5 Time integration

With respect to the time integration of the governing equations, and having in mind a fast and efficient algorithm, we advocate for an explicit type of time integrator. For simplicity, an explicit one-step two-stage Total Variation Diminishing Runge-Kutta (TVD-RK) scheme has been preferred [29–31,78]. This is described by the following time update equations from time step t^n to t^{n+1} as

$$\mathcal{U}_{a}^{\star} = \mathcal{U}_{a}^{n} + \Delta t \, \dot{\mathcal{U}}_{a}^{n} (\mathcal{U}_{a}^{n}, t^{n});$$

$$\mathcal{U}_{a}^{\star\star} = \mathcal{U}_{a}^{\star} + \Delta t \, \dot{\mathcal{U}}_{a}^{\star} (\mathcal{U}_{a}^{\star}, t^{n+1});$$

$$\mathcal{U}_{a}^{n+1} = \frac{1}{2} (\mathcal{U}_{a}^{n} + \mathcal{U}_{a}^{\star\star}).$$
(45)

In this manuscript, the geometry (spatial locations of the particles) is also updated through the above TVD-RK algorithm [29–31,83,84]. This results in a monolithic time integration procedure where the vector of particle unknowns $\{p_a, J_a, x_a\}$ (a = 1, ..., N, where N represents the total number of particles across the computational domain) are all updated via (45). The maximum time step Δt is governed by the standard Courant–Friedrichs–Lewy (CFL) condition given as

$$\Delta t = \alpha_{\rm CFL} \frac{h_{\rm min}}{c_{p,\rm max}},\tag{46}$$

where $c_{p,\text{max}}$ is the maximum *p*-wave speed, h_{\min} is the minimum particle spacing within the computational domain and α_{CFL} is the CFL stability number. For the numerical computations presented in this paper, a value of $\alpha_{\text{CFL}} = 0.3$, unless otherwise stated, has been chosen to ensure both accuracy and stability [29,31] of the algorithm.

Given the fact that the interacting force T_{ab} (30) is not co-linear with the position vector $(x_b - x_a)$ [54], the SPH algorithm described above (29a,b) does not intrinsically fulfil conservation of angular momentum. To address this issue, we adapt the global projection algorithm introduced by Lee et al. [31]. Specifically, the internal nodal force $\sum_{b \in \Lambda_a^b} T_{ab}$ and the particle based stabilisation terms, namely $\sum_{b \in \Lambda_a^b} \mathcal{D}_v^{ab}$ and $\sum_{b \in \Lambda_a^b} \mathcal{D}_{\Delta C}^{ab}$, described in (29a) are suitably modified in a least-square sense in order to preserve the total angular momentum, whilst still ensuring the global conservation of linear momentum.

6 Algorithmic description

For ease of understanding, Algorithm 1 summarises the complete algorithmic description of the mixed-based $\{p, J\}$ Upwind Smooth Particle Hydrodynamics (Upwind-SPH) methodology, with all the necessary numerical ingredients.

Algorithm 1: Complete stabilised Upwind-SPH mixed methodology					
Input : \mathcal{U}_a^n where $\mathcal{U} = [p, J]^T$ Output : \mathcal{U}_a^{n+1} , P_a^{n+1} , \mathbf{x}_a^{n+1}					
(1) ASSIGN old primary variables: $\mathcal{U}_a^{\text{old}} = \mathcal{U}_a^n$ and $\mathbf{x}_a^{\text{old}} = \mathbf{x}_a^n$					
(2) EVALUATE <i>p</i> -wave speed c_p (58)					
(3) COMPUTE time increment: Δt					
for <i>TVD-RK</i> time integrator = 1 to 2 do					
(4) COMPUTE right-hand-side of the mixed-based system: \dot{p}_a (29a) (with T_{ab} described in (30)) and \dot{J}_a (29b)					
(5) APPLY discrete angular momentum preserving algorithm (6) EVOLVE { \mathcal{U}_a, x_a } via TVD-RK (45)					
(7) COMPUTE first Piola P_a ((10)) end					
(8) UPDATE { $\mathcal{U}_{a}^{n+1}, \mathbf{x}_{a}^{n+1}$ } (see 45)					
(9) COMPUTE first Piola P_a^{n+1} (10)					

7 Numerical examples

In this section, a series of two dimensional (2D) and three dimensional (3D) numerical examples are presented in order to assess the robustness, effectiveness and applicability of the new SPH computational framework. Notice that as the numerical examples included in this paper are restricted to inviscid flow dynamics (and importantly, without severe changes in topology), strictly speaking, no physical dissipation, but only numerical, will be present in our simulations.

For post-processing purposes, one popular option is to plot the solutions directly based on the particle values. This however may not be viable when a fluid patch experiences extremely large deformation and the distance between particles can become excessively large, unless a prohibitively large number of particles is used in the simulation, which is not the most preferred option in industry. In order to circumvent this shortcoming, we present another simple option to visualise the results. This is achieved by introducing a secondary set of particles, with their values approximated using an appropriate kernel interpolation procedure [54] (ensuring zeroth- and first-order completeness). Advantages of this alternative visualisation procedure are demonstrated in the subsequent examples.

7.1 Wave-like cubical fluid patch

As it is well known, one of the persistent drawbacks of the SPH method is the difficulty to ensure a proper order of convergence of the overall algorithm. Spatial order of convergence is still regarded as one of the key challenges by SPH developers [4].

The aim of this example is to examine the spatial convergence of the proposed $\{p, J\}$ Upwind-SPH methodology for the unknown fields velocity and pressure. A three dimensional cube of length L = 2 m, as shown in Fig. 3a, is analysed with symmetric boundary conditions for all the surfaces. When small deformations are considered, this example has a closed-form solution for the velocity field described as

$$\boldsymbol{v}(\boldsymbol{X}, t) = \omega A_0 \cos(\omega t) \boldsymbol{V};$$

$$\boldsymbol{V} = \begin{bmatrix} A \sin\left(\frac{\pi X}{L}\right) \cos\left(\frac{\pi Y}{L}\right) \cos\left(\frac{\pi Z}{L}\right) \\ B \cos\left(\frac{\pi X}{L}\right) \sin\left(\frac{\pi Y}{L}\right) \cos\left(\frac{\pi Z}{L}\right) \\ C \cos\left(\frac{\pi X}{L}\right) \cos\left(\frac{\pi Y}{L}\right) \sin\left(\frac{\pi Z}{L}\right) \end{bmatrix},$$
(47)

where $\omega = 2\pi \text{ s}^{-1}$ and $A_0 = 5 \times 10^{-4} \text{ m}$. The parameters $\{A, B, C\}$ are set as A = -2, B = C = 1 to ensure the existence of an incompressible flow. The problem is initialised with a zero pressure field (see Fig. 3b) together with an initial velocity field defined by using t = 0 in (47) (see Fig. 3c). For the fluid patch to be in equilibrium, a specific representation of the body force term is required

$$f(X,t) = -\omega^2 A_0 \sin(\omega t) V.$$
(48)

An elastic fluid type of constitutive law is used with density $\rho_0 = 1000 \text{ kg/m}^3$, bulk modulus $\kappa = 10 \text{ MPa}$ and $\gamma = 1$. As presented in 58, the initial pressure wave speed used in this case is $c_p^0 = \sqrt{\frac{\kappa \gamma}{\rho_0}} = 100 \text{ m/s}.$



Fig. 3 Wave-like cubical fluid patch: problem setup. (Color figure online)

Convergence analysis is carried out by measuring both the L_1 and L_2 error norms between the numerical and analytical solutions. Fig. 4a, b show equal second-order convergence for both the velocity field and the pressure field, which is one of the main advantages of the proposed SPH methodology. In this case, mesh refinement improves at the same rate results for both velocity and pressure.

7.2 Stretching of a circular fluid patch

This two dimensional example was first proposed by Monaghan [9], and later explored by several authors in References [54,92,98,99]. The main objective of this example is to verify the accuracy of the developed SPH algorithm and to show the preservation of total linear momentum. In this example, analytical expressions for the velocity (or displacement) and pressure fields are provided. Their detailed derivations can be found in [9,54,99].

The circular fluid patch has a radius *R* of 1 m and is free from any external forces, as illustrated in Fig. 5a. Both the initial velocity field v_0 and the initial pressure field p_0 are, respectively, described as follows

$$\boldsymbol{v}_0 = \begin{bmatrix} -A_0 X\\ A_0 Y\\ 0 \end{bmatrix}; \qquad p_0 = \frac{1}{2}\rho_0 A_0^2 \left[R^2 - \left(X^2 + Y^2 \right) \right], \tag{49}$$

where $A_0 = 100 \text{ s}^{-1}$. For visualisation purposes, contour plots of the above initial conditions are also displayed in Fig. 5b, c. The fluid properties used in this example are density $\rho_0 = 1000 \text{ kg/m}^3$ and bulk modulus $\kappa = 1.96 \text{ GPa}$. Utilising the value of $\gamma = 1$ in the equation of state, the initial pressure wave speed of the fluid patch can be computed as $c_p^0 := \sqrt{\frac{\kappa\gamma}{\rho_0}} = 1400 \text{ m/s}$. The domain is spatially discretised with a uniformly distributed particle arrangement. Moreover, the outermost radial particles are set free in order to allow the boundary surface to deform freely.

We start off by benchmarking the positions of the particles, situated at the highest and lowest points along the semi-major axis, against the published results reported in References [9,54]. Due to the symmetric nature of the problem, the solutions of those two particles are found to be identical, and hence only the results based on the highest particle are presented (see Table 1). It can be seen that the results obtained with the proposed method agree extremely well with those published. The maximum difference with respect to the analytical solution is calculated to be 0.75%.

To verify the convergence of the problem, a particle refinement study is examined (see Fig. 6). Particles are sequentially refined in order to compare the resolution of the deformed fluid patch at $tA_0 = 1.294$. Notice that the particles remain stable (without any oscillations) whilst being stretched. The free surface is accurately captured and no particle clumping is observed. The clumping of SPH particles has been reported in Reference [99] when the particles are distributed along the axes of a Cartesian coordinate system. This shortcoming can also be circumvented by introducing either a boundary conforming particle distribution [54] or a priori particle redistribution algorithm [99].

It can be seen that the pressure plot based on particle values is no longer visible (refer to the top row of Fig. 6) when the fluid patch experiences a very large stretch. For this reason, the visualisation is carried out with the help of a secondary set of particles with values interpolated via SPH shape functions (24c). As shown in the bottom row of Fig. 6, the deformed shapes of the fluid patch are practically identical. Pressure resolution is further enhanced when using a larger number of particles. For qualitative comparison purposes, Table 2 shows the positions of the semi-minor ($X = [1, 0, 0]^T$ m)





Fig.4 Wave-like cubical fluid patch: **a** L_1 and (2) L_2 norm convergence of components of pressure, *X*-velocity, *Y*-velocity and *Z*-velocity using the proposed { p, *J* } Upwind-SPH scheme. Results are obtained with A = -2, B = C = 1, $\omega = 2\pi$ s⁻¹ and $A_0 = 5 \times 10^{-4}$ m at time

t = 0.9375 ms. An elastic fluid is used with density $\rho_0 = 1000$ kg/m³, bulk modulus $\kappa = 10$ MPa, material coefficient $\gamma = 1$ and $\Delta t = 9.375 \times 10^{-5}$ s



Fig. 5 Stretching of a circular fluid patch: problem setup

Table 1Stretching of a circularfluid patch: comparison ofposition for the highest particlealong the semi-major axis fromanalytical, standard SPH,corrected SPH (CSPH) and $\{p, J\}$ Upwind-SPH

Time (s)	Analytical	Standard SPH ¹	CSPH ²	$\{p, J\}$ Upwind-SPH
0.0008	1.081	1.086	_	1.086 (0.46%)
0.0020	1.219	1.220	1.220	1.222 (0.25%)
0.0038	1.438	1.440	—	1.440 (0.14%)
0.0050	1.595	1.610	1.600	1.604 (0.56%)
0.0076	1.943	1.910	—	1.957 (0.72%)
0.0080	2.000	2.020	2.000	2.015 (0.75%)

Percentage differences with respect to the analytical solution are stated within the brackets

¹ Refer to Ref. [9]

² Refer to Ref. [54] (only three sets of results were reported)



Fig. 6 Stretching of a circular fluid patch: Particle refinement of deformed fluid patch alongside with pressure distribution plotted as particles (top) and contour (bottom) at $tA_0 = 1.294$ using of **a** 317, **b** 1257, **c** 5025, and **d** 20,081 particles. Red dashed lines represent the ana-

lytical deformation. Results obtained using {p, J} Upwind-SPH with an initial velocity and pressure (refer to Eq. 49). An elastic fluid of density $\rho_0 = 1000 \text{ kg/m}^3$, bulk modulus $\kappa = 1.96 \text{ GPa}$, material coefficient $\gamma = 1$ and $\alpha_{\text{CFL}} = 0.3$.

Table 2 Stretching of a circular fluid patch: comparison of position for farthest particle along the $(X = [1, 0, 0]^T \text{ m})$ and semi-major $(X = [0, 1, 0]^T \text{ m})$ axes between sequential particle density using $\{p, J\}$ Upwind-SPH against analytical solution at $tA_0 = 1.294$

Particle refinement No. of particles	P1 317	P2 1257	P3 5025	P4 20,081	Analytical
Semi-minor axis (m)	0.3806 (2.04%)	0.378 (1.34%)	0.377 (1.07%)	0.376 (0.80%)	0.373
Semi-major axis (m)	2.723 (1.38%)	2.713 (1.01%)	2.709 (0.86%)	2.698 (0.45%)	2.686

The percentage difference against analytical solution for each particle density is stated in bracket





Fig. 7 Time evolution of **a** numerical dissipation using a discretisation of 317, 709, 1257 and 2821 particles; and **b** components of linear

using {p, J} Upwind-SPH with an initial velocity and pressure (refer to Eq. 49). An elastic fluid of density $\rho_0 = 1000 \text{ kg/m}^3$, bulk modulus $\kappa = 1.96 \text{ GPa}$, material coefficient $\gamma = 1$ and $\alpha_{\text{CFL}} = 0.3$

and semi-major ($X = [0, 1, 0]^{T}$ m) axes at various time instants. As expected, the obtained results converge to the available analytical solution.

momentum using a discretisation of 20,081 particles. Results obtained

Figure 7a shows the time evolution of the accumulated numerical entropy (dissipation) present in the algorithm. This is achieved by integrating the Hamiltonian energy of the system described in (35) over time. The total numerical dissipation is reduced when successively increasing the particle density. Furthermore, Fig. 7b depicts the time history of the components of global linear momentum of the system. As expected, the global linear momentum fluctuates around zero machine accuracy in the absence of external forces.

The predicted solution is expected to oscillate around and then gradually converge towards the incompressible flow solution. The incompressibility limit is governed by the bulk modulus of the fluid, which in turn affects the corresponding pressure wave speed. Notice here that an increase in the bulk modulus of the fluid patch would lead to a smaller time increment used in the simulation. To assess the effectiveness of the algorithm in the near incompressibility regime, the pressure evolution is monitored at position $X = [0, 0, 0]^{T}$ using three different values of the bulk modulus, namely κ , 4κ and 16κ . As shown in Fig. 8, all the results converge towards the incompressible solution without the appearance of non-physical pressure instabilities.

For completeness, Fig. 9 shows a sequence of snapshots and their pressure contours. Majority of the authors [9,54,92,98,99] presented the deformation up to $tA_0 = 2$. To demonstrate the stability of the algorithm in a long-term response, we continue running the simulation up to $tA_0 = 4$. Two particles, one located at the semi-major axis and the other one located at the semi-minor axis, are monitored and compared against the analytical solutions (see Table 3). The particles at both axes are stretched to approximately 6.5 times and 2.9 times of their original positions.

7.3 Rotation of a square fluid patch

Similar to the example presented in Sect. 7.2, a square patch of inviscid fluid without the imposition of any external forces is considered. This problem was first proposed by Colagrossi [14], and this has since been used as an interesting benchmark validation example in the SPH community [68,100,101]. This problem is particularly challenging as spurious zero energy modes[8,92] would accumulate and, eventually, lead to the breakdown of the scheme. One viable option to remove this instability is the use of kernel derivatives that is fixed in



Fig. 8 Stretching of a circular fluid patch: time evolution of nondimensional pressure at $X = [0, 0, 0]^{T}$ m with varying bulk modulus κ . Results obtained using 1257 particles with an initial velocity and

pressure (refer to Eq. 49). An elastic fluid of density $\rho_0 = 1000 \text{ kg/m}^3$, material coefficient $\gamma = 1$ and $\alpha_{CFL} = 0.3$



Fig. 9 Stretching of a circular fluid patch: Evolution of the deformation alongside with pressure distribution using $\{p, J\}$ Upwind-SPH at (left to right) $tA_0 = \{0.0, 0.228, 0.421, 0.592, 0.747, 1.019, 1.141, 1.367, 1.472, 1.759, 2.015, 2.467, 3.042, 3.536, 4.0\}$. Results obtained using

1257 particles with an initial velocity and pressure (refer to Eq. 49). An elastic fluid of density $\rho_0 = 1000 \text{ kg/m}^3$, bulk modulus $\kappa = 1.96 \text{ GPa}$, material coefficient $\gamma = 1$ and $\alpha_{\text{CFL}} = 0.3$.

Table 3	Stretching of a circular fluid	patch: Time evolution of farthest	particle along the semi-maj	jor and semi-minor axes usin	$g\{p, J\}$ Upwind-SPH

tA_0 (-)	0.0	0.228	0.421	0.592	0.747	1.019	1.141	1.367	1.472	1.759	2.015	2.467	3.042	3.536	4.0
A-Mjr. (m)	1.0	1.251	1.492	1.717	1.927	2.302	2.472	2.787	2.935	3.339	3.699	4.338	5.150	5.848	6.489
Mjr. (m)	1.0	1.253	1.497	1.725	1.940	2.322	2.495	2.816	2.966	3.376	3.742	4.390	5.212	5.917	6.563
Diff. (%)	0.0	0.16	0.34	0.47	0.67	0.87	0.93	1.04	1.06	1.11	1.16	1.20	1.20	1.18	1.14
A-Mnr. (m)	1.0	0.799	0.670	0.583	0.519	0.434	0.405	0.359	0.341	0.300	0.270	0.231	0.194	0.171	0.154
Mnr. (m)	1.0	0.800	0.673	0.587	0.524	0.440	0.410	0.364	0.346	0.305	0.275	0.235	0.198	0.174	0.157
Diff. (%)	0.0	0.15	0.43	0.72	0.94	1.24	1.33	1.51	1.58	1.70	1.78	1.87	1.91	1.87	2.01

A-represents the analytical solution. Mjr. and Mnr. refer to semi-major and semi-minor axes respectively



Fig. 10 Rotation of a square fluid patch: problem setup

Fig. 11 Rotation of a square fluid patch: Particle refinement of deformed fluid patch alongside with pressure distribution plotted at particles (top) and as a contour plot (bottom) at $tA_0 = 1.1$ using **a** 441, **b** 1681, **c** 6561 and **d** 25,921 particles. Grey dashed line represents the initial position, black dashed lines are the trajectories and red dashed line represents the free surface deformation from ANSYS Fluent using an ultrafine reference mesh. Results obtained using $\{p, J\}$ Upwind-SPH with an initial velocity and pressure distributions (Refer to Eq. 50). An elastic fluid of density $\rho_0 = 1000 \, \text{kg/m}^3$, bulk modulus $\kappa = 1.96$ GPa, material coefficient $\gamma = 1$ and $\alpha_{\rm CFL} = 0.3$





Fig. 12 Rotation of a square fluid patch: time evolution of **a** trajectory of the corner particle at the material point $X = [-0.5, 0.5, 0]^{T}$ m; and **b** numerical dissipation using 441, 1681, 6561 and 25,921 particles.



Fig. 13 Rotation of a square fluid patch: time evolution of components a global linear momentum; and b global angular momentum using a discretisation of 6561 particles. Results obtained using $\{p, J\}$ Upwind-

the reference configuration (inherent to the use of Lagrangian SPH formalism [41]), which is exactly one of the key motivations of this work.

The fluid patch, of length L = 1 m, is subject to an initial velocity and pressure distribution described as

$$\boldsymbol{v}_{0} = \begin{bmatrix} A_{0}Y\\ -A_{0}X\\ 0 \end{bmatrix};$$
$$p_{0} = \rho_{0}\sum_{m}^{\infty}\sum_{n}^{\infty} -\frac{32\omega^{2}}{\left[mn\pi^{2}\left(\frac{m\pi}{L}\right)^{2} + \left(\frac{n\pi}{L}\right)^{2}\right]}$$



Results obtained using {p, J} Upwind-SPH with an initial velocity and pressure (refer to Eq. 50). An elastic fluid of density $\rho_0 = 1000 \text{ kg/m}^3$, bulk modulus $\kappa = 1.96 \text{ GPa}$, material coefficient $\gamma = 1$ and $\alpha_{\text{CFL}} = 0.3$



SPH with an initial velocity and pressure (refer to Eq. 50). An elastic fluid of density $\rho_0 = 1000 \text{ kg/m}^3$, bulk modulus $\kappa = 1.96 \text{ GPa}$, material coefficient $\gamma = 1$ and $\alpha_{\text{CFL}} = 0.3$

$$\sin\left(\frac{m\pi X^*}{L}\right)\sin\left(\frac{n\pi Y^*}{L}\right),\tag{50}$$

where $A_0 = 200 \text{ s}^{-1}$, $X^* = X + \frac{L}{2}$ and $Y^* = Y + \frac{L}{2}$. The parameters *m* and *n* are odd numbers and the pressure field (50b) converges rapidly when $\{m, n\} \ge 3$. The centre of mass of this fluid patch is located at $X = [0, 0, 0]^{\text{T}}$ m and the initial density is taken as $\rho_0 = 1000 \text{ kg/m}^3$. Using both the bulk modulus of $\kappa = 1.96$ GPa and the coefficient of $\gamma = 1$, the pressure wave speed at time t = 0 becomes $c_p^0 = 1400$ m/s.

By imposing zero pressure gradient at the four corners of the fluid patch, the particles initially located at these cor-

Fig. 14 Rotation of a square fluid patch: deformed fluid patch at $tA_0 = 1.1$ using **a** {**p**, J} Upwind-SPH with F ensures zero-th order completeness and **b** $\{p, J\}$ Upwind-SPH with Fensures zero-th and first-order completeness. Results obtained using 6561 particles with an initial velocity and pressure (refer to Eq. 50). An elastic fluid of density $\rho_0 = 1000 \, \text{kg/m}^3$, bulk modulus $\kappa = 1.96$ GPa, material coefficient $\gamma = 1$ and $\alpha_{\rm CFL} = 0.3.$ (a) $\{p, J\}$ Upwind-SPH with F (31) approximated (b) $\{p, J\}$ Upwind-SPH with F (31) approximated without correction C_{ab} with correction $\tilde{\mathbf{C}}_{ab}$. -0.1 -0.05 -0.15 0 $p/(\rho_0 A_0^2 L^2)$ Fig. 15 Rotation of a square fluid patch: Deformed fluid patch using $\mathbf{a} \{ \mathbf{p}, J \}$ Upwind-SPH without dissipation terms (e.g. $\mathcal{D}_{v}^{ab} = \mathcal{D}_{\Delta C}^{ab} = \mathbf{0} \text{ and } \mathcal{D}_{p}^{ab} = 0)$ and $\mathbf{b} \{p, J\}$ Upwind-SPH at $tA_0 = 0.324$. Results obtained using 6561 particles with an initial velocity and pressure (refer to Eq. 50). An elastic fluid of density $\rho_0 = 1000 \,\text{kg/m}^3$, bulk modulus $\kappa = 1.96$ GPa, material coefficient $\gamma = 1$ and $\alpha_{CFL} = 0.3$ (a) $\{p, J\}$ Upwind-SPH with $\mathcal{D}_{v}^{ab} = \mathcal{D}_{\Delta C}^{ab} = \mathbf{0}$ (b) $\{p, J\}$ Upwind-SPH and $\mathcal{D}_{p}^{ab} = 0$ -0.15 -0.1 -0.05 0 $p/(\rho_0 A_0^2 L^2)$ -0.2 -0.1 0 -0.3

 $p/\left(\rho_{0}A_{0}^{2}L^{2}
ight)$

Fig. 16 Rotation of a square fluid patch: evolution of the deformation alongside with pressure distribution using {p, J} Upwind-SPH at (left to right) $tA_0 = \{0.085, 0.268, 0.594, 0.903, 1.466, 2.416, 3.570\}$.

Results obtained using 6561 particles with an initial velocity and pressure (refer to Eq. 50). An elastic fluid of density $\rho_0 = 1000 \text{ kg/m}^3$, bulk modulus $\kappa = 1.96 \text{ GPa}$, material coefficient $\gamma = 1$ and $\alpha_{\text{CFL}} = 0.3$



Fig. 17 Swirling of a square fluid patch: geometry configuration and initial conditions

ners would evolve in time in the direction of the velocity field [102]. To examine this, a particle refinement analysis at time $tA_0 = 1.1$ is presented in Fig. 11. Both the free surface deformation and the pressure representation for every level of particle refinement are displayed. For benchmarking purposes, we simulate the exact same problem using the commercial software package ANSYS Fluent via an ultrafine discretisation of 2 million elements, which can then be treated as a reference solution. The proposed method accurately captures the location of the corners, as well as the bending of four arms, even with the utilisation of very few particles. No spurious mechanism is observed, which typically appears between $tA_0 = 0.6$ and $tA_0 = 1.2$ as reported in [68,102].

Figure 12a monitors the trajectory path of the particle at $X = [-0.5, 0.5, 0]^{T}$ m. After a relatively long time, the position of the particle becomes inaccurate due to the accumulation of the numerical dissipation. This is commonly found in the work of the other authors [14,103] who attempted to simulate this example via a stabilisation procedure. However, dramatic improvement is seen with increased particle density. For completeness, the total numerical dissipation of the algorithm is measured. Figure 12b shows a clear reduction in the numerical dissipation by increasing the level of refinement.

Figure 13a, b show the time history of the components of the global linear and angular momenta of the system. As expected, the angular momentum components are perfectly conserved and the linear momentum components fluctuate around zero machine accuracy. Crucially, Fig. 14 shows the necessity for accurate evaluation of the deformation gradient F. Wrong deformation path is observed when employing the standard kernel gradient approximation C_{ab} [8] that only ensures zeroth-order consistency. Such inaccurate deformation can be rectified by utilising the correction technique to evaluate the deformation F (31) (refer to Fig. 14b). Moreover, Fig. 15 highlights the importance of adding sufficient amount of numerical dissipation to the algorithm. As displayed in Fig. 15a, four corners of the fluid patch exhibit spurious mechanisms when using the {p, J} SPH algorithm without introducing any sort of stabilisation terms (i.e. set $\mathcal{D}_{v}^{ab} = \mathcal{D}_{\Delta C}^{ab} = 0$ and $\mathcal{D}_{p}^{ab} = 0$). This shortcoming can be alleviated using suitable dissipation terms (see Fig. 15b), consistently derived from the satisfaction of global Hamiltonian energy.

For visualisation purposes, Fig. 16 shows a sequence of deformed states for the rotating fluid patch. Given the initial conditions described in (50), the fluid patch initially is dominated by a centrifugal force [99]. This would allow the fluid patch first to rotate and then to start developing the fluid arms at those four corners. The arms would continue to elongate and deform, and eventually creating four extremely thin bent arms at a later time. Very smooth pressure field is observed.

7.4 Swirling of a square fluid patch

Using exactly the same geometry as shown in Fig. 10a, another type of square fluid patch is assessed [98,102,104]. In this case, the fluid patch is subject to a specific representation for the velocity field described as follows

$$\boldsymbol{v}_{0} = -A_{0}L \begin{bmatrix} e^{-4} - e^{-16(Y/L)^{2}} \\ e^{-4} + e^{-16(X/L)^{2}} \\ 0 \end{bmatrix},$$
(51)

where $A_0 = 100 \text{ s}^{-1}$. This particular type of velocity distribution enables the fluid patch to undergo extremely massive distortion. The adopted fluid properties are material density $\rho_0 = 1000 \text{ kg/m}^3$, bulk modulus $\kappa = 1.96 \text{ GPa}$ and the fluid coefficient of $\gamma = 1$.

At time t = 0, the fluid domain is subjected to a pair of rotating vorticity fields pushing the fluid patch to deform



Fig. 18 Swirling of a square fluid patch: particle refinement of deformed fluid patch alongside with pressure distribution plotted as particles (top) and contour (bottom) at $tA_0 = 0.2$ using a 441, b 1681, c 6561 and d 25,921 particles. Grey dashed line represents the initial position and red dashed line refers to the free surface deformation from

ANSYS Fluent. Results obtained using $\{p, J\}$ Upwind-SPH with an initial velocity and pressure distributions (refer to Eq. 51). An elastic fluid of density $\rho_0 = 1000 \text{ kg/m}^3$, bulk modulus $\kappa = 1.96 \text{ GPa}$, material coefficient $\gamma = 1$ and $\alpha_{\text{CFL}} = 0.3$. (Color figure online)





Fig. 19 Time evolution of **a** relative deformation of the nondimensional corner particle at the material point $X = [0.5, 0.5, 0]^{T}$ m with its position shown in ' \Box ' up until $tA_0 = 0.5$; and **b** numerical dissipation 441, 1681, 6561 and 25,921 particles. Results obtained

using {**p**, J} Upwind-SPH with an initial velocity and pressure (refer to Eq. 51). An elastic fluid of density $\rho_0 = 1000 \text{ kg/m}^3$, bulk modulus $\kappa = 1.96 \text{ GPa}$, material coefficient $\gamma = 1$ and $\alpha_{\text{CFL}} = 0.3$

Fig. 20 Swirling of a square fluid patch: Deformed fluid patch at $tA_0 = 0.5$ using **a** $\{p, J\}$ Upwind-SPH with Fensures only zeroth order completeness and $\mathbf{b} \{ p, J \}$ Upwind-SPH with F ensures both zeroth- and first-order completeness (31) Upwind-SPH. Results obtained using 6561 particles with an initial velocity and pressure distributions (refer to Eq. 51). An elastic fluid of density $\rho_0 = 1000 \, \text{kg/m}^3$, bulk modulus $\kappa = 1.96 \,\text{GPa}$, material coefficient $\gamma = 1$ and $\alpha_{\rm CFL} = 0.3$



(a) $\{p, J\}$ Upwind-SPH with F approximated without gradient correction \mathbf{C}_{ab} (b) $\{p, J\}$ Upwind-SPH with F approximated with gradient correction $\tilde{\mathbf{C}}_{ab}$ (31)

1	-0.5	0	0.5	
	p/	$(\rho_0 A_0^2)$	$L^{2})$	

along the flow direction. The two corner particles located at the centre of the vortices remain always at rest. The remaining corners of the domain are located at the flow symmetry direction, and hence known as free boundary particles. In order to show the convergence of the overall algorithm, a particle refinement study is carried out at time $tA_0 = 0.2$ and is shown in Fig. 18. Both the top and left surfaces of the fluid patch are being pushed towards the domain, whereas all the corner particles remain at rest. As time evolves, the free boundary particle located at position $X = [-0.5, -0.5, 0]^T$ m (i.e. the right bottom corner of the domain) gradually moves away from its original location. Insofar as the symmetric vorticity field is considered, the boundary particle propagates in a linear fashion. The movement evolution relative to its original position is depicted in Fig. 19a. Interestingly, the particle travels linearly with respect to its initial position for all particle refinements. The position of the particle converges with increasing particle density, which can be observed through the location of markers. Moreover, significant reduction in the amount of numerical dissipation is observed in Fig. 19b. This indeed shows that the numerical dissipation term is consistent and would vanish by increasing the number of particles.

Figure 20 demonstrates the importance of the proper evaluation of the deformation F. Clearly, the deformation path is wrongly captured without imposing any kernel gradient cor-



Fig. 21 Swirling of a square fluid patch: Deformed fluid patch using **a** {**p**, J} Upwind-SPH by setting $\mathcal{D}_{v}^{ab} = \mathcal{D}_{\Delta C}^{ab} = \mathbf{0}$ and $\mathcal{D}_{p}^{ab} = 0$, **b** {**p**, J} Upwind-SPH by setting $\mathcal{D}_{p}^{ab} = 0$ and **c** {**p**, J} Upwind-SPH

at $tA_0 = 0.256$. Results obtained using 6561 particles with an initial velocity and pressure distributions (refer to Eq. 51). An elastic fluid of density $\rho_0 = 1000 \text{ kg/m}^3$, bulk modulus $\kappa = 1.96 \text{ GPa}$, material coefficient $\gamma = 1$ and $\alpha_{\text{CFL}} = 0.3$



rection (refer to Fig. 20a) unless excessively fine meshes are used. As shown in Fig. 20b, this inaccurate deformed shape is corrected by using the enhanced kernel gradient approximation. It is well known that, sufficient amount of numerical stabilisation that needs to be added to the algorithm remains a persistent numerical issue in the SPH community. The SPH algorithm clearly exhibits pressure fluctuations without the introduction of any sort of stabilisation by setting the values of $\mathcal{D}_v^{ab} = \mathcal{D}_{\Delta C}^{ab} = \mathbf{0}$ and $\mathcal{D}_p^{ab} = 0$. Addition of \mathcal{D}_v^{ab} and $\mathcal{D}_{\Delta C}^{ab}$ to the linear momentum evolution alone would partially alleviate the mechansims at the beginning of the simulation Fig. 21b, c. The mechanism however would accumulate and re-emerge at a later time. As displayed in Fig. 22, this mode can be entirely removed by using the proposed SPH algorithm where $\{\mathcal{D}_{v}^{ab}, \mathcal{D}_{p}^{ab}, \mathcal{D}_{\Delta C}^{ab}\}$ are appropriately introduced without the need to tune any user defined stabilisation parameters.

Finally, Fig. 23 shows a series of snapshots for the fluid patch, displaying the deformation patterns and the pressure field. A very smooth and stable pressure pattern is observed.



Fig. 23 Swirling of a square fluid patch: evolution of the deformation alongside with pressure distribution using $\{p, J\}$ Upwind-SPH at (left to right) $tA_0 = \{0.0, 0.20, 0.51, 0.80, 1.01, 1.30\}$. Results obtained

using 25,921 particles with an initial velocity and pressure (refer to Eq. 51). An elastic fluid of density $\rho_0 = 1000 \text{ kg/m}^3$, bulk modulus $\kappa = 1.96 \text{ GPa}$, material coefficient $\gamma = 1$ and $\alpha_{\text{CFL}} = 0.3$



Fig. 25 Collapsing water column: particle refinement of deformed fluid patch alongside with pressure distribution plotted as contour at $t\sqrt{g/L_0} = 0.9$ using a 1331, b 9261 and c 68,921 particles. Red dashed line represents the free surface deformation from ANSYS Fluent and the square markers represent the free surface solutions from Kelecy and

Pletcher [105]. Results obtained using {p, J} Upwind-SPH with an initial hydrostatic pressure. An elastic fluid of density $\rho_0 = 1000 \text{ kg/m}^3$, bulk modulus $\kappa = 16 \text{ MPa}$, material coefficient $\gamma = 7$ and $\alpha_{\text{CFL}} = 0.3$. (Color figure online)

7.5 Collapsing water column

In this section, a classical three dimensional benchmark example is presented (see Fig. 24). A cubic water column, of length L = 0.05715 m, in hydrostatic equilibrium is confined between a wall and a gate. The oscillating water column is subject to a gravitational acceleration of g = 9.81 m/s² acting vertically downwards. The material properties under consideration are density $\rho_0 = 1000$ kg/m³, bulk modulus $\kappa = 16$ kPa and fluid coefficient $\gamma = 7$. Symmetric boundary conditions are imposed on the solid walls.

In this example, the gate is removed instantaneously and the fluid column collapses under the gravitational force. Specifically, a number of experimental and numerical results are available in the literature for benchmarking purposes. Figure 25 shows a refinement analysis at a particular time $t\sqrt{g/L_0} = 0.9$, with special emphasis on the deformation patterns of the free surface alongside pressure distribution. With respect to the free surface deformation, the proposed upwind SPH algorithm agrees well with other published results (e.g. Kelecy and Pletcher [105] and ANSYS Fluent). Furthermore, the time varying positions at both the surge front and the column height of the wall are monitored and compared [2,54,105–110] in Fig. 26a, b. Again, very good agreement is observed even with very few particles.

Essentially, an appropriate evaluation for the deformation map F prevents an unrealistic collapsing trend of the water column (refer to Fig. 27). Most of the particles belonging to the surge regions are seen to be clumped together and some are seen to be protruding across the bottom wall. In addition, a dip is observed at column height.

Figure 28 illustrates the deformation history of the water column, displaying pressure contour. As already reported in [47,54,80], pressure instabilities would be excited without incorporating any numerical damping to the Jacobian evolution (see Fig. 28a,b). This indeed can be eliminated by introducing a consistent Riemann-based dissipation term \mathcal{D}_p^{ab} (refer to Fig. 28c), without affecting the order of convergence of the algorithm. Another popular option to address this drawback is to employ the δ -SPH method [46,67], and which requires a user defined stabilisation parameter for the evaluation of the Laplacian operator.

7.6 Impact of a fluid patch on a wall

The final example of this work is the numerical analysis of an inviscid fluid patch impacting on a wall. The setup of this problem is very similar to the work reported in References [67,68,77] when considering impact of two identical fluid patches. A cuboid, with a height of H = 1.5 m (or L = 1 m in Fig. 29a) and a unit cross section, is subject to an initial



Fig. 26 Collapsing water column: time evolution of **a** surge front and **b** column height using 1331, 9261 and 68,921 particles. Markers represent experimental results and dashed lines represent numerical solutions. Results obtained using {p, J} Upwind-SPH with an initial hydrostatic pressure. An elastic fluid of density $\rho_0 = 1000 \text{ kg/m}^3$, bulk modulus $\kappa = 16 \text{ MPa}$, material coefficient $\gamma = 7$ and $\alpha_{CFL} = 0.3$

dropping velocity field

$$\boldsymbol{v}_0 = A_0 L \begin{bmatrix} 0\\-1\\0 \end{bmatrix},\tag{52}$$

where $A_0 = 1 \text{ s}^{-1}$. The fluid properties used in this example are density $\rho_0 = 1000 \text{ kg/m}^3$, bulk modulus $\kappa = 10 \text{ MPa}$ and the fluid coefficient $\gamma = 1$, which in turn yields the corresponding pressure wave speed $c_p^0 = 100 \text{ m/s}$. Moreover, symmetric boundary conditions are enforced at the bottom surface, whereas the remaining boundary surfaces are set free.

We begin this example by performing a particle refinement analysis at time $tA_0 = 0.098$ (see Fig. 30). The deformed shape of the free surface predicted by the proposed SPH method agrees extremely well with the ANSYS Fluent solutions via an ultrafine discretisation of 1.6 million elements. Pressure resolution is improved by increasing the number



(a) $\{p, J\}$ Upwind-SPH with F approximated without gradient correction



Fig. 27 Collapsing water column: deformed fluid patch at
$$t\sqrt{g/L_0} =$$
 1.965 using **a** {*p*, *J*} Upwind-SPH with *F* only ensure zeroth-order completeness and **b** {*p*, *J*} Upwind-SPH with *F* ensures both zeroth-and first-order completeness (31). Results obtained using 9261 parti-

cles with an initial hydrostatic pressure field. An elastic fluid of density $\rho_0 = 1000 \text{ kg/m}^3$, bulk modulus $\kappa = 16 \text{ MPa}$, material coefficient $\gamma = 7$ and $\alpha_{\text{CFL}} = 0.3$

taneously generated and then followed by a tensile wave.

These waves would then propagate towards the top free sur-

face and reflect back to the fluid domain. A smooth pressure

of particles. The results are further analysed by monitoring the displacement at two different positions, one at a surge front $X = [0.5, 0, 0]^T$ m (see Fig. 31a) and the other one at a column height $X = [0, 1.5, 0]^T$ m (see Fig. 31b). It is seen that the time history of the horizontal displacement at the surge front location converges to the reference solution (i.e. ANSYS Fluent) when increasing the number of particles. More interestingly, the vertical displacement at the column height is converged even with the use of a small number of particles, showing optimal convergence of the algorithm. Very small oscillations in column height being observed due to the nature of the compressible algorithm.

Figure 32 depicts a series of snapshots of the pressure evolution simulated using the proposed Upwind-SPH method. These snapshots are captured at every time interval of $\Delta t A_0 = 0.001$ up until $t A_0 = 0.03$. After the fluid patch is in contact with the wall, a compressive shock wave is instan-

distribution is observed throughout the evolution without the need to introduce extra artificial (or non-consistent) viscosity into the algorithm. To check if the method suffers from long-term instability, we continue running the simulation for a longer time. As shown in Fig. 33, extremely good results are obtained via a Riemann based SPH algorithm. To further examine the robustness of the algorithm, we simulate the exact same problem but this time applied on a

simulate the exact same problem but this time applied on a hemisphere. Figure 34 shows the importance of adding sufficient amount of numerical dissipation to the volume map conservation law. The proposed upwind SPH algorithm performs extremely well in this case without any numerical difficulties.



Fig. 28 Collapsing water column: evolution of the deformation (top to bottom) alongside pressure distribution using **a** {p, J} Upwind-SPH by setting $\mathcal{D}_{v}^{ab} = \mathcal{D}_{\Delta C}^{ab} = \mathbf{0}$ and $\mathcal{D}_{p}^{ab} = 0$, **b** {p, J} Upwind-SPH

with $\mathcal{D}_p^{ab} = 0$ and **c** {p, J} Upwind-SPH at $t\sqrt{g/L_0} = \{0.39, 0.59, 0.98, 1.46, 1.96, 2.49\}$. Results obtained using 9261 particles with an hydrostatic pressure. An elastic fluid of density $\rho_0 = 1000 \text{ kg/m}^3$, bulk modulus $\kappa = 16 \text{ MPa}$, material coefficient $\gamma = 7$ and $\alpha_{\text{CFL}} = 0.3$

8 Conclusions

In this paper, a new Total Lagrangian Smooth Particle Hydrodynamics computational framework has been introduced for the numerical analysis of inviscid fluid flows. The methodology is established starting from a mixed-based system of first-order hyperbolic conservation laws, where the linear momentum p conservation law is solved along with the conservation equation for the Jacobian of the deformation *J*. For problems without experiencing severe changes in topological surface, the use of a Total Lagrangian description removes the need for continuous (updated) particle neighbouring search (either removing it altogether or delaying it until an extremely large number of computational time steps has taken place), yielding an extremely efficient scheme still capable of handling massive deformations.



Fig. 29 Impact of a cuboid fluid patch on a wall: problem setup

Fig. 30 Impact of a cuboid fluid patch on a wall: particle refinement of deformed fluid patch alongside with pressure distribution plotted as contour at $tA_0 = 0.098$ using **a** 2501, **b** 9801, c 38,801 and d 154,401 particles. Red dashed line represents the free surface deformation from ANSYS Fluent. Results obtained using $\{p, J\}$ Upwind-SPH with an initial velocity $v_0 = A_0[0, -1, 0]^{T}$ m/s using $A_0 = 1 s^{-1}$ and pressure $p_0 = 0$ Pa. An elastic fluid of density $\rho_0 = 1000 \text{ kg/m}^3$, bulk modulus $\kappa = 10$ MPa, material coefficient $\gamma = 1$ and $\alpha_{\text{CFL}} = 0.3$. (Color figure online)







Fig. 31 Impact of a cuboid fluid patch on a wall: time evolution of **a** surge front and **b** column height using 2501, 9801, 38,801 and 154,401 particles. Black dashed line represents the numerical solutions for validation purposes. Results obtained using $\{p, J\}$ Upwind-SPH with an

initial velocity $\mathbf{v}_0 = A_0[0, -1, 0]^{\mathrm{T}}$ m/s using $A_0 = 1$ s⁻¹ and pressure $p_0 = 0$ Pa. An elastic fluid of density $\rho_0 = 1000$ kg/m³, bulk modulus $\kappa = 10$ MPa, material coefficient $\gamma = 1$ and $\alpha_{\mathrm{CFL}} = 0.3$

From the spatial discretisation point of view, an entropystable Smooth Particle Hydrodynamics is presented where consistent Riemann-based (upwinding) numerical dissipation is suitably introduced guaranteeing global numerical entropy production, the latter demonstrated by monitoring the so-called Hamiltonian energy of the system. Such numerical dissipation is very well designed from a mathematical standpoint (taking into consideration of the characteristic structure of the hyperbolic system) and, crucially, does not rely on the use of any user defined artificial stabilisation parameters. From the temporal discretisation point of view, an explicit one-step two-stage Total Variation Diminishing Runge Kutta time integrator combined with an angular momentum preserving algorithm is presented.

Finally, a comprehensive list of challenging prototype problems has been presented in order to benchmark the results obtained against alternative numerical strategies available in the literature, including the standard SPH [8,32,40], the Corrected SPH [54,55] and ANSYS Fluent [110]. It has been shown that the resulting SPH algorithm overcomes a number of numerical difficulties posed by SPH methods when simulating slightly compressible fluids, namely non-physical hydrostatic pressure fluctuations, hour-glassing and numerical errors associated with global conservation, completeness, long-term instability and convergence. Very importantly, both velocities and pressure (or the volumetric strain) display equal second order of convergence. As a result, the new SPH algorithm provides a good balance between accuracy and speed of computation.

The next step of our work is the adaptation of the current Total Lagrangian SPH algorithm to handle violent freesurface flows typically accompanied with large topological changes, by incorporating alternative Updated Lagrangian formalism in the style of [28].

Acknowledgements The first author would like to acknowledge the financial support of the Welsh Government, the European Regional Development Fund (ERDF) and the funding provided via the ASTUTE 2020 operation. The second author gratefully acknowledges the support provided by the EPSRC Strategic Support Package: Engineering of Active Materials by Multiscale/Multiphysics Computational Mechanics-EP/R008531/1. The third author acknowledges the financial support received through the European Training Network Protection (Project ID: 764636).

Compliance with ethical standards

Conflicts of interest On behalf of all authors, the corresponding author states that there is no conflict of interest.

Open Access This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecomm ons.org/licenses/by/4.0/.

Appendix A Hyperbolicity

This Appendix is included to demonstrate the hyperbolicity [85,86] of system (8). In addition, the computation of the



Fig. 32 Impact of a cuboid fluid patch on a wall: time evolution of pressure monitored at material point $X = [0, 0, 0]^{T}$ m. The blue dots represent the time instant (every 0.001 s) of the deformation alongside with pressure distribution (left to right; top to bottom) using 102,541 particles. Results obtained using $\{p, J\}$ Upwind-SPH with an initial

velocity $\mathbf{v}_0 = A_0[0, -1, 0]^{\mathrm{T}}$ m/s using $A_0 = 1 \ s^{-1}$ and pressure $p_0 = 0$ Pa. An elastic fluid of density $\rho_0 = 1000 \text{ kg/m}^3$, bulk modulus $\kappa = 10$ MPa, material coefficient $\gamma = 1$ and $\alpha_{\text{CFL}} = 0.3$. (Color figure online)



Fig. 33 Impact of a cuboid fluid patch on a wall: evolution of the deformation (top to bottom) alongside with pressure distribution using a $\{p, J\}$ Upwind-SPH with $\mathcal{D}_{v}^{ab} = \mathcal{D}_{\Delta C}^{ab} = 0$ and $\mathcal{D}_{p}^{ab} = 0$, b $\{p, J\}$ Upwind-SPH with $\mathcal{D}_{p}^{ab} = 0$ and c $\{p, J\}$ Upwind-SPH at

 $tA_0 = \{0.00487, 0.0355, 0.1355, 0.255, 0.3672, 0.549, 0.7935\}.$ Results obtained using 102,541 particles with an initial velocity $v_0 = A_0[0, -1, 0]^{\text{T}}$ m/s using $A_0 = 1 \text{ s}^{-1}$ and pressure $p_0 = 0$ Pa. An elastic fluid of density $\rho_0 = 1000 \text{ kg/m}^3$, bulk modulus $\kappa = 10$ MPa, material coefficient $\gamma = 1$ and $\alpha_{\text{CFL}} = 0.3$

Computational Particle Mechanics



0

 $p/\left(\rho_0 c_p^0 A_0 L\right)$

-0.5

0.5

1

(a) { \boldsymbol{p}, J } Upwind-SPH with $\boldsymbol{\mathcal{D}}_{\boldsymbol{v}}^{ab}$ $\boldsymbol{\mathcal{D}}_{\Delta \mathbf{C}}^{ab} = \mathbf{0}$ and $\mathcal{D}_{p}^{ab} = 0$ (b) $\{\boldsymbol{p}, J\}$ Upwind-SPH with $\mathcal{D}_p^{ab} = 0$ =

-1

(c) $\{\boldsymbol{p}, J\}$ Upwind-SPH

Fig. 34 Impact of a hemisphere fluid patch on a wall: evolution of the deformation (top to bottom) alongside with pressure distribution using **a** {**p**, J} Upwind-SPH with $\mathcal{D}_{v}^{ab} = \mathcal{D}_{\Delta C}^{ab} = \mathbf{0}$ and $\mathcal{D}_{p}^{ab} = 0$, **b** {**p**, J} Upwind-SPH with $\mathcal{D}_{p}^{ab} = 0$ and **c** {**p**, J} Upwind-SPH at $tA_{0} =$

using $A_0 = 1 \text{ s}^{-1}$ and pressure $p_0 = 0$ Pa. An elastic fluid of density $\rho_0 = 1000 \text{ kg/m}^3$, bulk modulus $\kappa = 10 \text{ MPa}$, material coefficient $\gamma = 1$ and $\alpha_{CFL} = 0.3$

{0.003, 0.0607, 0.1310, 0.22, 0.303, 0.395, 0.48}. Results obtained

using 136,393 particles with an initial velocity $v_0 = A_0[0, -1, 0]^{T}$ m/s

maximum (pressure) wave speed is necessary for the evaluation of the maximum time step of the explicit time integrator. This can be achieved by, first recalling that P = -pH (10) and, combining Eq. (6) multiplied by $H^{Ave}N$ with Eq. (4) to give

$$U^{2} = -\frac{\Lambda_{A}^{2}}{\rho_{0}} \frac{\llbracket p \rrbracket}{\llbracket J \rrbracket}; \qquad \Lambda_{A}^{2} = \left(\boldsymbol{H}^{\text{Ave}}\boldsymbol{N}\right) \cdot \left(\boldsymbol{H}^{\text{Ave}}\boldsymbol{N}\right).$$
(53)

In this paper, we consider the simple case of an "elastic fluid" type of model described in (10), that is $p = \kappa (J^{-\gamma} - 1)$. In this case

 $\llbracket p \rrbracket = \kappa \llbracket J^{-\gamma} \rrbracket.$ (54)

In the case of slightly compressible fluid, it is instructive to consider small jumps in $[J^{-\gamma}]$ by approximating it using a Taylor series expansion as

$$[J^{-\gamma}] \approx -\frac{\gamma [\![J]\!]}{(J^{\text{Ave}})^{\gamma+1}}; \qquad J^{\text{Ave}} = \frac{1}{2}(J^+ + J^-).$$
 (55)

Combining (55) with the second line of (54) would give an alternative expression for the jump in pressure described as

$$\llbracket p \rrbracket = -\frac{\kappa \gamma}{(J^{\text{Ave}})^{\gamma+1}} \llbracket J \rrbracket.$$
(56)

Finally, the determination of pressure wave speed can now follow by substituting Eqs. (56) into (53), which results in

$$U = \pm c_p; \qquad c_p = \Lambda_A \sqrt{\frac{\gamma \kappa}{\rho_0 (J^{\text{Ave}})^{\gamma+1}}}.$$
(57)

It is worth noticing that expression above (57) degenerates to the so-called linear acoustic pressure wave speed c_p^{Lin} by evaluating the nonlinear pressure wave speed c_p (57) at the initial undeformed configuration (that is, H = I and J = 1)

$$c_p^{\rm Lin} = \sqrt{\frac{\gamma\kappa}{\rho_0}}.$$
(58)

References

- 1. Floryan JM, Rasmussen H (1989) Numerical methods for viscous flows with moving boundaries. Appl Mech Rev 42(12):323–341. https://doi.org/10.1115/1.3152416
- Tsai W, Yue DKP (1996) Computation of nonlinear free-surface flows. Annu Rev Fluid Mech 28(1):249–278. https://doi.org/10. 1146/annurev.fl.28.010196.001341
- Scardovelli R, Zaleski S (1999) Direct numerical simulation of free-surface and interfacial flow. Annu Rev Fluid Mech 31(1):567–603. https://doi.org/10.1146/annurev.fluid.31.1.567
- Shadloo MS, Oger G, Le Touzé D (2016) Smoothed Particle Hydrodynamics method for fluid flows, towards industrial applications: motivations, current state, and challenges. Comput Fluids 136:11–34. https://doi.org/10.1016/j.compfluid.2016.05.029
- Liu MB, Liu GR, Lam KY (2006) Adaptive Smooth Particle Hydrodynamics for high strain hydrodynamics with material strength. Shock Waves 15(1):21–29. https://doi.org/10.1007/ s00193-005-0002-1
- Lucy LB (1977) A numerical approach to the testing of the fission hypothesis. Astron J 82:1013–1024. https://doi.org/10.1086/ 112164
- Gingold RA, Monaghan JJ (1977) Smoothed particle hydrodynamics—theory and application to non-spherical stars. Mon Not R Astron Soc 181:375–389. https://doi.org/10. 1093/mnras/181.3.375
- Monaghan JJ (1985) Particle methods for hydrodynamics. Comput Phys Rep 3(2):71–124. https://doi.org/10.1016/0167-7977(85)90010-3
- Monaghan JJ (1994) Simulating free surface flows with SPH. J Comput Phys 110(2):399–406. https://doi.org/10.1006/jcph. 1994.1034
- Gomez-Gesteira M, Rogers BD, Dalrymple RA, Crespo AJC (2010) State-of-the-art of classical SPH for free-surface flows. J Hydraul Res 48:6–27. https://doi.org/10.1080/00221686.2010. 9641242
- Violeau D, Rogers BD (2016) Smooth Particle Hydrodynamics (SPH) for free-surface flows: past, present and future. J Hydraul Res 54(1):1–26. https://doi.org/10.1080/00221686. 2015.1119209
- Gotoh H, Khayyer A (2018) On the state-of-the-art of particle methods for coastal and ocean engineering. Coast Eng J 60(1):79– 103. https://doi.org/10.1080/21664250.2018.1436243
- Monaghan JJ, Kocharyan A (1995) SPH simulation of multi-phase flow. Comput Phys Commun 87(1):225–235. https://doi.org/10. 1016/0010-4655(94)00174-Z

- Colagrossi A, Landrini M (2003) Numerical simulation of interfacial flows by smooth particle hydrodynamics. J Comput Phys 191(2):448–475. https://doi.org/10.1016/S0021-9991(03)00324-3
- Hu XY, Adams NA (2007) An incompressible multi-phase SPH method. J Comput Phys 227(1):264–278. https://doi.org/10.1016/ j.jcp.2007.07.013
- Grenier N, Antuono M, Colagrossi A, Le Touzé D, Alessandrini B (2009) An Hamiltonian interface SPH formulation for multifluid and free surface flows. J Comput Phys 228(22):8380–8393. https://doi.org/10.1016/j.jcp.2009.08.009
- Luo M, Koh CG, Gao M, Bai W (2015) A particle method for twophase flows with large density difference. Int J Numer Methods Eng 103(4):235–255. https://doi.org/10.1002/nme.4884
- Luo M, Koh CG, Bai W, Gao M (2016) A particle method for twophase flows with compressible air pocket. Int J Numer Methods Eng 108(7):695–721. https://doi.org/10.1002/nme.5230
- Ellero M, Kröger M, Hess S (2002) Viscoelastic flows studied by smoothed particle dynamics. J Nonnewton Fluid Mech 105(1):35– 51. https://doi.org/10.1016/S0377-0257(02)00059-9
- Vázquez-Quesada A, Ellero M, Español P (2012) A SPH-based particle model for computational microrheology. Microfluid Nanofluid 13(2):249–260. https://doi.org/10.1007/s10404-012-0954-2
- Xu X, Ouyang J, Yang BX, Liu ZJ (2013) SPH simulations of three-dimensional non-Newtonian free surface flows. Comput Methods Appl Mech Eng 256:101–116. https://doi.org/10.1016/ j.cma.2012.12.017
- Price DJ (2012) Smooth particle hydrodynamics and magnetohydrodynamics. J Comput Phys 231(3):759–794. https://doi.org/10. 1016/j.jcp.2010.12.011
- Randles PW, Libersky LD (1996) Smoothed particle hydrodynamics: some recent improvements and applications. Comput Methods Appl Mech Eng 139:375–408. https://doi.org/10.1016/ S0045-7825(96)01090-0
- Belytschko T, Krongauz Y, Organ D, Fleming M, Krysl P (1996) Meshless methods: an overview and recent developments. Comput Methods Appl Mech Eng 139:3–47. https://doi.org/10.1016/ S0045-7825(96)01078-X
- Johnson GR, Beissel SR (1996) Normalized smoothing functions for SPH impact computations. Int J Numer Methods Eng 39:2725–2741. https://doi.org/10.1002/(SICI)1097-0207(19960830)39:16<2725::AID-NME973>3.0.CO;2-9
- Gray JP, Monaghan JJ, Swift RP (2001) SPH elastic dynamics. Comput Methods Appl Mech Eng 190:6641–6662. https://doi. org/10.1016/S0045-7825(01)00254-7
- Liu MB, Liu GR, Lam KY (2003) Constructing smoothing functions in smoothed particle hydrodynamics with applications. J Comput Appl Math 155(2):263–284. https://doi.org/10.1016/ S0377-0427(02)00869-5
- Vidal Y, Bonet J, Huerta A (2006) Stabilized updated Lagrangian corrected SPH for explicit dynamic problems. Int J Numer Methods Eng 69:2687–2710. https://doi.org/10.1002/nme.1859
- Lee CH, Gil AJ, Greto G, Kulasegaram S, Bonet J (2016) A new Jameson–Schmidt–Turkel smooth particle hdrodynamics algorithm for large strain explicit fast dynamics. Comput Methods Appl Mech Eng 311:71–111. https://doi.org/10.1016/j.cma.2016. 07.033
- Lee CH, Gil AJ, Hassan OI, Bonet J, Kulasegaram S (2017) A variationally consistent Streamline Upwind Petrov Galerkin Smooth Particle Hydrodynamics algorithm for large strain solid dynamics. Comput Methods Appl Mech Eng 318:514–536. https://doi. org/10.1016/j.cma.2017.02.002
- 31. Lee CH, Gil AJ, Ghavamian A, Bonet J (2019) A total Lagrangian upwind smooth particle hdrodynamics algorithm for large strain

explicit solid dynamics. Comput Methods Appl Mech Eng 344:209–250. https://doi.org/10.1016/j.cma.2018.09.033

- Monaghan JJ (1988) An introduction to SPH. Comput Phys Commun 48(1):89–96. https://doi.org/10.1016/0010-4655(88)90026-4
- Benz W (1990) Smoothed particle hydrodynamics: a review. Springer, Dordrecht, pp 269–288. https://doi.org/10.1007/978-94-009-0519-1_16
- Monaghan JJ (1992) Smooth particle hydrodynamics. Ann Rev Astron Astrophys 30(1):543–574. https://doi.org/10.1146/ annurev.aa.30.090192.002551
- Monaghan JJ (2005) Smoothed particle hydrodynamics. Rep Prog Phys 68:1703–1759. https://doi.org/10.1088/0034-4885/ 68/8/R01
- Liu MB, Liu GR (2010) Smoothed particle hydrodynamics (SPH): an overview and recent developments. Archiv Comput Methods Eng 17(1):25–76. https://doi.org/10.1007/s11831-010-9040-7
- 37. Monaghan JJ (2012) Smoothed Particle Hydrodynamics and its diverse applications. Annu Rev Fluid Mech 44(1):323–346. https://doi.org/10.1146/annurev-fluid-120710-101220
- Dyka CT, Ingel RP (1995) An approach for tension instability in smoothed Particle hydrodynamics (SPH). Comput Struct 57:573– 580. https://doi.org/10.1016/0045-7949(95)00059-P
- Dyka CT, Randles PW, Ingel RP (1998) Stress points for tension instability in SPH. Int J Numer Methods Eng 40(13):2325– 2341. https://doi.org/10.1002/(SICI)1097-0207(19970715)40: 13<2325::AID-NME161>3.0.CO;2-8
- Monaghan JJ (2000) SPH without a tensile instability. J Comput Phys 159:290–311. https://doi.org/10.1006/jcph.2000.6439
- Bonet J, Kulasegaram S (2001) Remarks on tension instability of Eulerian and Lagrangian corrected smooth particle hydrodynamics (CSPH) methods. Int J Numer Methods Eng 52:1203–1220. https://doi.org/10.1002/nme.242
- 42. Swegle JW, Hicks DL, Attaway SW (1995) Smooth particle hydrodynamics stability analysis. J Comput Phys 116:123–134. https://doi.org/10.1006/jcph.1995.1010
- Belytschko T, Guo Y, Liu WK, Xiao SP (2000) A unified stability analysis of meshless particle methods. Int J Numer Methods Eng 48:1359–1400. https://doi.org/10.1002/ 1097-0207(20000730)48:9<1359::AID-NME829>3.0.CO;2-U
- Puso MA, Chen JS, Zywicz E, Elmer W (2008) Meshfree and finite element nodal integration methods. Int J Numer Methods Eng 74:416–446. https://doi.org/10.1002/nme.2181
- Molteni D, Colagrossi A (2009) A simple procedure to improve the pressure evaluation in hydrodynamic context using the SPH. Comput Phys Commun 180(6):861–872. https://doi.org/10.1016/ j.cpc.2008.12.004
- Antuono M, Colagrossi A, Marrone S, Molteni D (2010) Freesurface flows solved by means of SPH schemes with numerical diffusive terms. Comput Phys Commun 181(3):532–549. https:// doi.org/10.1016/j.cpc.2009.11.002
- 47. Bonet J, Burton AJ (1998) A simple average nodal pressure tetrahedral element for incompressible and nearly incompressible dynamic explicit applications. Commun Numer Methods Eng 14:437–449. https://doi.org/10.1002/(SICI)1099-0887(199805)14:5<437::AID-CNM162>3.0.CO;2-W
- Dohrmann CR, Heinstein MW, Jung J, Key SW, Witkowski WR (2000) Node-based uniform strain elements for three-node triangular and four-node tetrahedral meshes. Int J Numer Methods Eng 47:1549–1568. https://doi.org/10.1002/(SICI)1097-0207(20000330)47:9<1549::AID-NME842>3.0.CO;2-K
- Puso MA, Solberg J (2006) A stabilized nodally integrated tetrahedral. Int J Numer Methods Eng 67:841–867. https://doi.org/10. 1002/nme.1651

- Gee MW, Dohrmann CR, Key SW, Wall WA (2009) A uniform nodal strain tetrahedron with isochoric stabilization. Int J Numer Methods Eng 78:429–443. https://doi.org/10.1002/nme.2493
- Johnson GR, Stryk RA, Beissel SR (1996) SPH for high velocity impact computations. Comput Methods Appl Mech Eng 139(1):347–373. https://doi.org/10.1016/S0045-7825(96)01089-4
- Chen JK, Beraun JE, Carney TC (1999) A corrective smoothed particle method for boundary value problems in heat conduction. Int J Numer Methods Eng 46:231–252. https://doi.org/10.1002/ (SICI)1097-0207(19990920)46:2<231::AID-NME672>3.0.CO; 2-K
- Chen JK, Beraun JE (2000) A generalized smoothed particle hydrodynamics method for nonlinear dynamic problems. Comput Methods Appl Mech Eng 190(1):225–239. https://doi.org/10. 1016/S0045-7825(99)00422-3
- Bonet J, Lok TS (1999) Variational and momentum preservation aspects of smooth particle hydrodynamic formulations. Comput Methods Appl Mech Eng 180:97–115. https://doi.org/10.1016/ S0045-7825(99)00051-1
- 55. Bonet J, Kulasegaram S (2000) Correction and stabilization of Smooth Particle Hydrodynamics methods with applications in metal forming simulations. Int J Numer Methods Eng 47:1189–1214. https://doi.org/10.1002/(SICI)1097-0207(20000228)47:6<1189::AID-NME830>3.0.CO;2-I
- Bonet J, Kulasegaram S (2002) A simplified approach to enhance the performance of smooth particle hydrodynamics methods. Appl Math Comput 126(2):133–155. https://doi.org/10.1016/ S0096-3003(00)00143-0
- Dilts GA (1999) Moving-least-squares-particle hydrodynamics I. Consistency and stability. Int J Numer Methods Eng 44(8):1115– 1155. https://doi.org/10.1002/(SICI)1097-0207(19990320)44: 8<1115::AID-NME547>3.0.CO;2-L
- Dilts GA (2000) Moving least-squares particle hydrodynamics II: conservation and boundaries. Int J Numer Methods Eng 48(10):1503–1524. https://doi.org/10.1002/1097-0207(20000810)48:10<1503::AID-NME832>3.0.CO;2-D
- 59. Liu WK, Jun S, Sihling DT, Chen Y, Hao W (1999) Multiresolution reproducing kernel particle method for computational fluid dynamics. Int J Numer Methods Fluids 24(12):1391–1415. https://doi.org/10.1002/(SICI)1097-0363(199706)24:12<1391:: AID-FLD566>3.0.CO;2-2
- Liu WK, Jun YF, Zhang S (1995) Reproducing kernel particle methods. Int J Numer Methods Fluids 20(89):1081–1106. https:// doi.org/10.1002/fld.1650200824
- Belytschko T, Krongauz Y, Dolbow J, Gerlach C (1998) On the completeness of meshfree particle methods. Int J Numer Methods Eng 43(5):785–819. https://doi.org/10.1002/(SICI)1097-0207(19981115)43:5<785::AID-NME420>3.0.CO;2-9
- Gomez-Gesteira M, Rogers BD, Crespo AJC, Dalrymple RA, Narayanaswamy M, Dominguez JM (2012) SPHysics development of a free-surface fluid solver—part 1: theory and formulations. Comput Geosci 48:289–299. https://doi.org/10. 1016/j.cageo.2012.02.029
- Crespo AJC, Domínguez JM, Rogers BD, Gómez-Gesteira M, Longshaw S, Canelas R, Vacondio R, Barreiro A, García-Feal O (2015) DualSPHysics: open-source parallel CFD solver based on smooth particle hydrodynamics (SPH). Comput Phys Commun 187:204–216. https://doi.org/10.1016/j.cpc.2014.10.004
- Morris JP, Monaghan JJ (1997) A switch to reduce SPH viscosity. J Comput Phys 136(1):41–50. https://doi.org/10.1006/jcph.1997. 5690
- Monaghan JJ (1989) On the problem of penetration in particle methods. J Comput Phys 82(1):1–15. https://doi.org/10.1016/ 0021-9991(89)90032-6

- 66. Marrone S, Antuono M, Colagrossi A, Colicchio G, Le Touzé D, Graziani G (2011) δ-SPH model for simulating violent impact flow. Comput Methods Appl Mech Eng 200(13):1526–1542. https://doi.org/10.1016/j.cma.2010.12.016
- Antuono M, Marrone S, Colagrossi A, Bouscasse B (2015) Energy balance in the δ-SPH scheme. Comput Methods Appl Mech Eng 289:209–226. https://doi.org/10.1016/j.cma.2015.02.004
- 68. Sun PN, Colagrossi A, Marrone S, Zhang AM (2017) The δplus-SPH model: simple procedures for a further improvement of the SPH scheme. Comput Methods Appl Mech Eng 315(Supplement C):25–49. https://doi.org/10.1016/j.cma.2016.10.028
- Monaghan JJ (1997) SPH and Riemann solvers. J Comput Phys 136(2):298–307. https://doi.org/10.1006/jcph.1997.5732
- 70. Vila JP (1999) On particle weighted methods and smooth particle hydrodynamics. Mathe Models Methods Appl Sci 09(2):161–209. https://doi.org/10.1142/S0218202599000117
- Ben Moussa B, Vila J (2000) Convergence of SPH method for scalar nonlinear conservation laws. SIAM J Numer Anal 37(3):863–887. https://doi.org/10.1137/S0036142996307119
- Inutsuka S (2002) Reformulation of smooth particle hydrodynamics with Riemann solver. J Comput Phys 179(1):238–267. https:// doi.org/10.1006/jcph.2002.7053
- Ferrari A, Dumbser M, Toro EF, Armanini A (2008) A new stable version of the SPH method in Lagrangian coordinates. Commun Comput Phys 4(2):378–404
- 74. Ferrari A, Dumbser M, Toro EF, Armanini A (2009) A new 3D parallel SPH scheme for free surface flows. Comput Fluids 38(6):1203–1217. https://doi.org/10.1016/j.compfluid.2008. 11.012
- Avesani D, Dumbser M, Bellin A (2014) A new class of movingleast-squares WENO-SPH schemes. J Comput Phys 270:278– 299. https://doi.org/10.1016/j.jcp.2014.03.041
- Puri K, Ramachandran P (2014) Approximate Riemann solvers for the Godunov SPH (GSPH). J Comput Phys 270:432–458. https://doi.org/10.1016/j.jcp.2014.03.055
- Marrone S, Colagrossi A, Di Mascio A, Le Touzé D (2015) Prediction of energy losses in water impacts using incompressible and weakly compressible models. J Fluids Struct 54:802–822. https:// doi.org/10.1016/j.jfluidstructs.2015.01.014
- Lee CH, Gil AJ, Bonet J (2013) Development of a cell centred upwind finite volume algorithm for a new conservation law formulation in structural dynamics. Comput Struct 118:13–38. https:// doi.org/10.1016/j.compstruc.2012.12.008
- 79. Lee CH, Gil AJ, Bonet J (2014) Development of a stabilised Petrov–Galerkin formulation for conservation laws in Lagrangian fast solid dynamics. Comput Methods Appl Mech Eng 268:40–64. https://doi.org/10.1016/j.cma.2013.09.004
- Gil AJ, Lee CH, Bonet J, Aguirre M (2014) A stabilised Petrov-Galerkin formulation for linear tetrahedral elements in compressible, nearly incompressible and truly incompressible fast dynamics. Comput Methods Appl Mech Eng 276:659–690. https://doi.org/10.1016/j.cma.2014.04.006
- Karim IA, Lee CH, Gil AJ, Bonet J (2014) A two-step Taylor Galerkin formulation for fast dynamics. Eng Comput 31:366– 387. https://doi.org/10.1108/EC-12-2012-0319
- Aguirre M, Gil AJ, Bonet J, Lee CH (2015) An upwind vertex centred finite volume solver for Lagrangian solid dynamics. J Comput Phys 300:387–422. https://doi.org/10.1016/j.jcp.2015.07.029
- Haider J, Lee CH, Gil AJ, Bonet J (2017) A first order hyperbolic framework for large strain computational solid dynamics: an upwind cell centred total Lagrangian scheme. Int J Numer Methods Eng 109:407–456. https://doi.org/10.1002/nme.5293
- 84. Haider J, Lee CH, Gil AJ, Huerta A, Bonet J (2018) An upwind cell centred total Lagrangian finite volume algorithm for nearly incompressible explicit fast solid dynamic applications. Comput

Methods Appl Mech Eng 340:684–727. https://doi.org/10.1016/ j.cma.2018.06.010

- Gil AJ, Lee CH, Bonet J, Ortigosa R (2016) A first order hyperbolic framework for large strain computational solid dynamics. Part II: Total Lagrangian compressible, nearly incompressible and truly incompressible elasticity. Comput Methods Appl Mech Eng 300:146–181. https://doi.org/10.1016/j.cma.2015.11.010
- Bonet J, Gil AJ, Lee CH, Aguirre M (2015) A first order hyperbolic framework for large strain computational solid dynamics. Part I: Total Lagrangian isothermal Elasticity. Comput Methods Appl Mech Eng 283:689–732. https://doi.org/10.1016/j.cma.2014.09. 024
- Aguirre M, Gil AJ, Bonet J, Carreno AA (2014) A vertex centred Finite Volume Jameson-Scdmidt-Turkel (JST) algorithm for a mixed conservation formulation in solid Dynamics. J Comput Phys 259:672–699. https://doi.org/10.1016/j.jcp.2013.12.012
- Bonet J, Lee CH, Gil AJ, Ghavamian A (2021) A first order hyperbolic framework for large strain computational solid dynamics. Part III: Thermo-elasticity. Comput Methods Appl Mech Eng 373:1–53. https://doi.org/10.1016/j.cma.2020.113505
- Green MD, Peiró J (2018) Long duration SPH simulations of sloshing in tanks with a low fill ratio and high stretching. Comput Fluids 174:179–199. https://doi.org/10.1016/j.compfluid.2018. 07.006
- Bonet J, Kulasegaram S, Rodriguez-Paz M, Profit M (2004) Variational formulation for the smooth particle hydrodynamics (SPH) simulation of fluid and solid problems. Comput Methods Appl Mech Eng 193(12):1245–1256. https://doi.org/10.1016/j. cma.2003.12.018
- Gurtin M, Fried E, Anand L (2010) The mechanics and thermodynamics of continua. Cambridge University Press, Cambridge
- Vaughan GL, Healy TR, Bryan KR, Sneyd AD, Gorman RM (2008) Completeness, conservation and error in SPH for fluids. Int J Numer Methods Fluids 56(1):37–62. https://doi.org/10.1002/ fld.1530
- Feldman J, Bonet J (2007) Dynamic refinement and boundary contact forces in SPH with applications in fluid flow problems. Int J Numer Methods Eng 72:295–324. https://doi.org/10.1002/ nme.2010
- 94. Hassan OI, Ghavamian A, Lee CH, Gil AJ, Bonet J, Auricchio F (2019) An upwind vertex centred finite volume algorithm for nearly and truly incompressible explicit fast solid dynamic applications: total and Updated Lagrangian formulations. J Comput Phys 3:1–34. https://doi.org/10.1016/j.jcpx.2019.100025
- Kluth G, Despres K (2010) Discretization of hyperelasticity on unstructured mesh with a ell-centered Lagrangian scheme. J Comput Phys 229(24):9092–9118. https://doi.org/10.1016/j.jcp.2010. 08.024
- 96. Maire PH, Abgrall R, Breil J, Ovadia J (2007) A cell-centered Lagrangian scheme for two-dimensional compressible flow problems. SIAM J Sci Comput 29(4):1781–1824. https://doi.org/10. 1137/050633019
- Georges G, Breil J, Maire PH (2017) A 3D finite volume scheme for solving the updated Lagrangian form of hyperelasticity. Int J Numer Methods Fluids 84(1):41–54. https://doi.org/10.1002/fld. 4336
- Jiang T, Ouyang J, Ren JL, Yang BX, Xu XY (2012) A mixed corrected symmetric SPH (MC-SSPH) method for computational dynamic problems. Comput Phys Commun 183(1):50–62. https:// doi.org/10.1016/j.cpc.2011.08.016
- Colagrossi A, Bouscasse B, Antuono M, Marrone S (2012) Particle packing algorithm for SPH schemes. Comput Phys Commun 183(8):1641–1653. https://doi.org/10.1016/j.cpc.2012.02.032
- Oger G, Doring M, Alessandrini B, Ferrant P (2007) An improved SPH method: towards higher order convergence. J Comput Phys 22(2):1472–1492. https://doi.org/10.1016/j.jcp.2007.01.039

- Oger G, Marrone S, Le Touzé D, de Leffe M (2016) SPH accuracy improvement through the combination of a quasi-Lagrangian shifting transport velocity and consistent ALE formalisms. J Comput Phys 313(Supplement C):76–98. https://doi.org/10.1016/j.jcp.2016.02.039
- 102. Colagrossi A (2005) A meshless Lagrangian method for freesurface and interface flows with fragmentation. Ph.d. thesis, Università di Roma La Sapienza
- 103. Antuono M, Colagrossi A, Marrone S, Lugni C (2011) Propagation of gravity waves through an SPH scheme with numerical diffusive terms. Comput Phys Commun 182(4):866–877. https:// doi.org/10.1016/j.cpc.2010.12.012
- 104. Ren J, Jiang T, Lu W, Li G (2016) An improved parallel SPH approach to solve 3D transient generalized Newtonian free surface flows. Comput Phys Commun 205:87–105. https://doi.org/ 10.1016/j.cpc.2016.04.014
- Kelecy FJ, Pletcher RH (1997) The development of a free surface capturing approach for multidimensional free surface flows in closed containers. J Comput Phys 138(2):939–980. https://doi. org/10.1006/jcph.1997.5847

- 106. Martin JC, Moyce WJ (1952) Some gravity wave problems in the motion of perfect liquids—Part IV. An experimental study of the collapse of liquid columns on a rigid horizontal plane. Philos Trans R Soc Lond A Math Phys Eng Sci 244(882):312–324. https://doi. org/10.1098/rsta.1952.0006
- 107. Hu C, Sueyoshi M (2010) Numerical simulation and experiment on dam break problem. J Mar Sci Appl 9(2):109–114. https://doi. org/10.1007/s11804-010-9075-z
- 108. Sheu TWH, Yu CH, Chiu PH (2009) Development of a dispersively accurate conservative level set scheme for capturing interface in two-phase flows. J Comput Phys 228(3):661–686. https://doi.org/10.1016/j.jcp.2008.09.032
- 109. Yang L, Gil AJ, Carreno AA, Bonet J (2018) Unified one-fluid formulation for incompressible flexible solids and multiphase flows: application to hydrodynamics using the immersed structural potential method (ISPM). Int J Numer Methods Fluids 86(1):78–106. https://doi.org/10.1002/fld.4408
- 110. (2011) ANSYS Documentations. ANSYS Inc., release 17.2 edn

Publisher's Note Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.